

Received 2 May 2024, accepted 7 June 2024, date of publication 12 June 2024, date of current version 2 July 2024.

Digital Object Identifier 10.1109/ACCESS.2024.3413354

TOPICAL REVIEW

Multi-Port Non-Isolated DC–DC Converters and Their Control Techniques for the Applications of Renewable Energy

R. ARAVIND^{®1}, BHARATIRAJA CHOKKALINGAM^{®1}, (Senior Member, IEEE), RAJESH VERMA², SAKTHIVEL ARUCHAMY³, (Member, IEEE),

AND LUCIAN MIHET-POPA^{®4}, (Senior Member, IEEE)

¹Centre for Electric Mobility (CEM), Department of Electrical and Electronics Engineering, SRM Institute of Science and Technology, Kattankulathur, Chennai 603203 India

²Electrical Engineering Department, College of Engineering, King Khalid University, Abha, Asir 61411, Saudi Arabia

³Department of Electrical and Electronics Engineering, New Horizon College of Engineering, Bengaluru, Karnataka 560103, India

⁴Faculty of Information Technology, Engineering and Economics, Østfold University College, 1757 Halden, Norway

Corresponding authors: Bharatiraja Chokkalingam (bharatic@srmist.edu.in) and Lucian Mihet-Popa (lucian.mihet@hiof.no)

This work was supported by the Deanship of Research and Graduate Studies at King Khalid University through the Large Group Research Project under Grant RGP 2/57/45.

ABSTRACT In recent times, the increasing demand for energy consumption on a global scale and the prevalent use of power electronics DC-DC converters in various applications, such as hybrid energy systems, hybrid vehicles, aerospace, satellites, and portable electronic devices. To increase the converters' dependability, effectiveness, adaptability, and cost-effectiveness, they have undergone substantial research and development. Diverse input levels can be merged for diverse output levels using new converter topologies, which increase their possibilities from single input-single output setups to multi-input-multiple configurations. The goal of ongoing research is to lower costs and component counts while raising efficiency and dependability. This review paper focuses on the analysis of non-isolated converters, especially for SI-MIMO topologies produced from different converter types. The research work, also includes control techniques and methods like proportional integral derivative, sliding mode control, model predictive control, state-space modelling, fuzzy logic control, and maximum power point tracking, which are all integrated with non-isolated DC-DC converters while considering things like settling concerns, response time, and complexity.

INDEX TERMS DC-DC converter, multiport converter, voltage gain boost, reduction of switch voltage stress, bidirectional converter, unidirectional converter, ZVS, ZCS, soft-switching, control systems.

ABBREVIATI	ON	CSM	Current-Source-Mode.
AC	Alternate Current.	D	Derivative.
ACO	Ant Colony Optimization.	DC	Direct Current.
ALC	Actively Switched Inductor-Capacitor.	DCM	Discontinuons Conduction Mode.
ANN	Artificial Neural Network.	DDTM	Double-Duty-Triple-Mode.
ASPWM	Augmented Sinusoidal Pulse Width Modu-	DE	Differential Evolution.
	lation.	DF	Dynamic Freewheeling.
CCM	Continuous Conduction Mode.	DIDO	Dual Input Dual Output.
CCS-MPC	Continuous Control Set Model Predictive	DISO	Dual Input Single Output.
	Control.	DMPPT	Distributed Maximum Power Point Track
			ing.
The associate	editor coordinating the review of this manuscript and	EGUs	Energy Generation Units.

approving it for publication was Zhixiang Zou¹⁰. EMI Electromagnetic Interference.

© 2024 The Authors. This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License.

FMS	Energy Management Strategy						
FSUs	Energy Storage Units						
EV	Electric Vehicles						
EV	Fuel Cells						
FC ECS MDC	Fuel Cells. Finite Control States Set Model Dradiative						
FCS-MPC	Finite Control-States Set Model Predictive						
ET C	Control.						
FLC	Fuzzy Logic Control.						
FOVs	Field-Of-Views.						
GA	Genetic Algorithm.						
HC	Hysteretic control.						
HCM	Hybrid Conduction Mode.						
HIIA	Hankel Interaction Index Array.						
Ι	Integral.						
ICR	Incremental Resistance.						
IGBTs	Insulated-Gate Bipolar Transistor.						
LED	Light-Emitting Diode.						
LMI	Linear Matrix Inequality.						
MIMO	Multi Input Multi Output.						
MISO	Multi Input Single Output.						
MOSFET	Metal-Oxide-Semiconductor Field-Effect						
	Transistor.						
MPC	Model Predictive Control.						
MPP	Maximum Power Point.						
MPPT	Maximum Power Point Tracking.						
MR-MIMO	Multicell Reconfigurable Multi Input Multi						
	Output						
NVS	Normalised Voltage Stress						
OFC	Output Filter Components						
OVACC	Output Voltage-Aware Charge Control						
P	Proportional						
P&O	Perturb and Observe						
PEM	Proton Exchange Membrane						
PI	Proportional-Integral						
	Proportional Integral Derivative						
DM	Participation Matric						
	Patter Semiconductor Components						
rscs	Power Semiconductor Components.						
	Particle Swarm Optimization.						
PSO-PID	Parucie Swarm Optimization Proportional						
DV	Integral Derivative.						
PV	Photovoltaic.						
PWM	Pulse width Modulation.						
PZI	Piezoelectric Transducer.						
RES	Renewable Energy Sources.						
RGA	Relative Gain Array.						
SC	Soft Computing.						
SEPIC	Single-Ended Primary-Inductor Converter.						
SIDO	Single Input Dual Output.						
SI-DIDO	Single Inductor Dual Input Dual Output.						
SI-DISO	Single Inductor Dual Input Single Output.						
SI-SIDO	Single Inductor Single Input Dual Output.						
SIMO	Single Input Multi Output.						
SI-MIMO	Single Inductor Multi Input Multi Output.						
SISO	Single-Input Single-Output.						
SSOs	Sub Synchronous Oscillations.						
SMC	Sliding Mode Control.						
	-						

State Space Modeling.

TLC	Three-Level Converter.
TSS-HB	Three-State Switching Hybrid Boost.
UPS	Uninterruptible Power Supply.
V	Voltage.
VMC	Voltage Multiplier Circuit.
VR	Voltage-Ripple.
VSI	Voltage Source Inverter.
VSM	Voltage-Source-Mode.
ZQR	Z- Quasi Resonant.

I. INTRODUCTION

The growing demand for clean energy has prompted the use of renewable energy sources (RES) as a viable method for energy generation [1]. Advancements in renewable energy-based power systems, hybrid vehicles, aerospace systems, smart grids utilizing renewable energy, and portable devices have posed challenges in designing new DC-DC power conversion systems. Multiple ports with changing voltage levels are required in a wide range of industrial applications, including photovoltaics, electric cars, data centres, and personal computers. Multiple independent single-input single-output (SISO) converters are commonly used to ensure voltage regulation and power control across these ports. However, this approach results in a significant number of components, leading to high overall costs and a large system volume. To tackle this issue, two types of integrated multiport DC-DC converters have been introduced in [2], the multiplexed semiconductor devices and integrating N-port converters based on typical buck [3], boost [4], and buckboost [5], [6] converters were proposed. These integrated converters effectively reduce the quantity of semiconductor devices and inductors, thereby improving both cost efficiency and power density. In the past years, oil-based power plants have been utilized [8]. The massive consumption of oil became scarce and eventually developed generation of hybrid power systems could be operated along with oil. However, the efficiency of power plants was observed due to the contribution of scarcity of natural gas and crude oil. Besides that, the burning of fossil fuels led to serious negative impacts on the environment, highly abundant and this method became costly [7], [8], [9], [10], [11]. RES has emerged as a new generation method, with multiple converter topologies and control strategies, [12], [13], [14], [15], [16], [17] additionally it has less maintenance and moderate efficiency.

Fig. 1 depicts the power converter family, categorizing both isolated and non-isolated converters and demonstrating standard converter topologies. The operation of a closed-loop DC-DC converter system is shown in Fig 2 Power converter operation is demonstrated in conjunction with a control technique aimed at stabilizing both the load and the source simultaneously. Furthermore, this control technique employs a feedback control loop from the load side to compare it with a desired reference value, aiming to achieve the optimal operation of the DC-DC converter.

The upcoming systems will consist of multiple input energy sources integrated through multi-input power

SSM



FIGURE 1. Flowchart of converter family [18].



FIGURE 2. Closed loop DC-DC converter system.



FIGURE 3. Conventional single – Port DC – DC converter structure.

electronics converters are demonstrated in [19], [20], and [21]. These converters can accommodate various input sources and leverage their advantages to provide controlled output for diverse applications.

In the literature, two structures for DC-DC conversion have been documented. The conventional structure illustrated in Fig. 3 involves combining multiple sources at a distinct load and utilizing separate DC-DC conversion stages for



FIGURE 4. Multiport DC – DC converter structure [21].

each source, allowing for independent control of the converters. This particular structure is commonly employed in grid-connected systems and standalone hybrid energy systems [19]. However, the multiple stages of power conversion and the need for communication devices in this system result in higher converter costs. Moreover, the independent control of multiple converters adds complexity to the system. To overcome these drawbacks, a multiport structure is adopted, as illustrated in Fig. 4. The entire structure is treated as a single power converter in this system, combining several sources, and the power regulation is handled by controllers. These multiport power electronics converters were created for a variety of purposes, including hybrid energy systems, hybrid cars, satellite/aerospace applications, and uninterrupted power supply. They offer advantages such as a simple structure, conversion stages that have a minimal number, and fewer devices. The low terminal voltage of supercapacitors and their wide range of voltage variations during charging and discharging operations present a significant difficulty. Therefore, a bidirectional DC-DC converter with a high voltage conversion ratio is needed to connect the lower supercapacitor voltage to the greater DC-link voltage [22]. Such applications are fundamentally dependent on the steep voltage conversion ratio provided by continuous gain-based bidirectional DC-DC converters.

This article examines recent developments in trends within the field of MIMO DC-DC converter topologies applied to renewable energy system applications. The study explores the operation of power converters, incorporating control techniques to concurrently stabilize both the load and source. Additionally, it delves into the advantages, disadvantages, and applications of MIMO DC-DC converters.

This comprehensive investigation undertakes the following scholarly endeavours:

- A meticulous study of parameters, comparing nonisolated DC-DC converter topologies with exacting precision.
- An exhaustive comparative analysis delving into the intricacies of design considerations relevant to a

spectrum of multiport converters tailored for renewable energy applications.

- A comprehensive assessment discerning the strengths and limitations of various multiport converter configurations, thereby offering nuanced insights.
- A rigorous comparative examination elucidating the efficacy of diverse control techniques deployed within multiport converter systems, contributing to a deeper understanding of their operational efficacy.

The remaining sections of the paper are organized as follows: Section II introduces single input single output DC-DC converters. Section III presents multi-input non-isolated DC-DC converters. Section IV introduces multi-output non-isolated DC-DC converters. Section V discusses various control techniques. Section VI addresses challenges and future trends. Finally, Section VII provides the conclusion.

II. SINGLE INPUT SINGLE OUTPUT DC-DC CONVERTER

Currently, there is a significant rise in non-isolated converter topologies due to the increasing demands of modern applications. However, conventional non-isolated converter topologies have some drawbacks that can negatively impact their efficiency and reduce the lifespan of the equipment they are integrated with. As a result, there is a growing trend towards hybridizing converter topologies as a means of addressing these challenges. Hybridizing converter topologies involves combining different converter topologies based on their unique features and limitations. In the case of non-isolated converter topologies, some limitations affect the performance. For example, features large input and output ripples, discontinuous input and output current, higher switching stress, and a higher duty cycle ratio. However, there have been recent advancements in the development of hybrid non-isolated DC-DC converter topologies, as illustrated in Fig. 5.

Fig. 5 (a) shows a transformer-less double-duty-triple mode DC-DC converter topology with high gain [23]. This topology offers several advantages over traditional non-isolated converter topologies, including a reduced component count and enhanced efficiency. Operating on two distinct duty cycles mitigates the challenges associated with a high duty cycle ratio. The converter operates in three different modes, leveraging power electronics components efficiently to achieve higher voltage gain and a stable output. Consequently, it eliminates the need for complex circuit structures such as voltage lift techniques, voltage multiplier circuits, and coupled transformers. Experimental testing has confirmed its suitability for DC-Microgrid applications, making it a viable integration option within renewable energy systems. The various topologies of Single Input Single Output converters are compared, and various applications are discussed in Table 1.

Its implementation facilitates smooth operation and enhances the lifespan of renewable energy extraction equipment, such as photovoltaic (PV) cells. A different converter configuration called the high gain three-state switching hybrid boost converter is introduced in [24], as depicted



FIGURE 5. (a) High gain transformer-less double-duty-triple-mode DC/DC converter. (b) High gain three-state switching hybrid boost converter.

in Fig. 5 (b). This converter belongs to the category of non-isolated topologies and shares similarities with the abovementioned hybrid converters. However, it incorporates an additional voltage lift circuit to achieve high voltage gain. count, enhanced output voltage gain, and optimal switching duty cycle ratios. By employing two distinct duty cycle ratios and operating in three different modes, it achieves excellent characteristics of voltage boosting. This topology is particularly suitable for voltage-boosting applications generated through renewable energy, such as fuel cells and, photovoltaic (PV) ensuring optimal efficiency. The control of single-input, single-output systems is covered in [25]. Switched Inductor Boost Converter has the following key features: high voltage conversion along with voltage-stress reduction across power electronic switch [26]. The inductor and capacitor in LC2D configuration are used to improve the current and voltage, respectively, while the diode protects the reverse flow of current [27]. A specific equation provides the closed-loop system's transfer function. Routh-Hurwitz's criterion is now less important from the standpoint of assessing the overall stability. A closed-loop system's transient response is determined by its closed-loop poles. A plot of the system poles when this gain varies from zero to infinity is called a root locus. A physical system experiences a transient after being abruptly activated by a sinusoidal waveform before reaching a steady state, also known as the sinusoidal steady state. The chapter uses both Nyquist and Bode charts to assess how close a system is to instability. A novel high-gain boost

Reference	Year	Туре	No. of Component	Applications			
[23]	2019	DC-DC	11	 Double-duty-triple-mode (DDTM) converter for dc-microgrid. Utilized for low input power 			
[24]	2019	DC-DC, DC-AC	11	 High gain three state switching hybrid boost (TSS-HB) converter for dc-microgrid. Single duty pulse confines the gain of voltage. 			
[28]	2018	DC-AC	13	 An ideal dc-dc converter for distributed generation has a high step-up and two coupled inductors. Elimination of turn off voltage spike. Transformerless grid connected PV systems 			
[29]	2016	DC-DC	13	 Stand-alone and grid-connected systems for renewable energy sources. Achieving high voltage levels without the need for magnetic components. 			
[30]	2021	DC-DC	12	 Anti-interference capability, high voltage gain, low component stress. Converter is appropriate for fuel cell vehicles. 			
[31]	2023	DC-DC	11	 Converter achieves significantly higher gain voltage factor. Single duty ratio perseveres with implication of converter. Suitable for high power applications 			

TABLE 1. Comparison of various topologies based on single input single output converters.

converter with a tri-switching state, which is non-isolated, has been proposed in [32]. Furthermore, it offers the advantages of reduced component The various optimization goals and constraints of battery energy storage systems are examined, with a specific emphasis on factors such as cost, capacity, lifespan, and emissions are discussed in [33].

III. MULTI INPUT NON – ISOLATED DC-DC CONVERTER

A development of multiport converters topologies can be stated. The converters have multiple input ports which have multiple power supplies. Furthermore, the multiport structures extended the opportunity of making the entire system simpler and more compact. This multi-input converter is further classified into DISO, DIDO & MISO.

A. DUAL INPUT SINGLE OUTPUT DC-DC CONVERTERS

A dual-input, single-output DC-DC converter is a versatile power electronics device designed to accept power from two different sources, offering flexibility and redundancy in power supply arrangements. These converters automatically or manually switch between inputs and are frequently employed in energy harvesting applications, which combine conventional power sources with gathered energy from sources like solar panels. High efficiency, small size, and dependability are the converters' main selling points, which make them ideal for a range of uses including IoT systems, renewable energy sets, and portable devices. A soft-switched DISO converter topology that can assimilate two low-voltage energy sources into a DC bus. Coupled inductor and voltage multiplier circuit (VMC) techniques have both been used to increase the voltage gain, as was presented in [34]. Electromagnetic interference (EMI) is significantly reduced due to the input and output ports sharing a common ground. Power can be transferred to the load from two separate input voltage sources. However, [36] Z- Quasi-Resonant (ZQR) converters are the most prominent features with individual energy glide manipulates among the sources along with easy design, implementation process and higher gain at a low duty cycle [34], [35], [36], [37], [38], [39], [40], [41], [42], [43]. The primary benefits of the converter listed in Table 2 include the need for fewer components, the possibility to use a variety of sources with various voltage and current characteristics, and the capacity to obtain high voltage gain without using high-frequency switching.

In the literature, several multi-input single-output DC-DC topologies have been published [38], [39], [40], [41]. The majority, however, use a buck or buck-boost gain relations for the output voltage. Recently, other topologies of the boost type that have a dual input and a single output have also been proposed. However, they have the following serious flaws, which restrict their uses for PV. The DISO DC-DC converter [41] offers several significant advantages, including the ability to achieve a high output voltage with a relatively low working duty ratio, continuous source current at both inputs, a shared ground, and reduced switch voltage stresses, among others.

DISO is given to the battery and used to charge the EV. The reliability of the drive is increased along with a reduction in the dependency on a single source when many sources are integrated to meet the needs of the load. Instead of a cascaded topology or a parallel topology, the direct integration of sources contributes [44], with a single inductor to reduce the cost and volume. The comparison of various dual input single output converter topologies is discussed in Table 3.

B. DUAL INPUT DUAL OUTPUT DC-DC CONVERTERS

Two input sources and two loads are integrated by the converter's four IGBTs, two inductors, two filter capacitors, and



 TABLE 2.
 SI-DISO converters is derived from Buck, Boost, Buck-Boost.

three diodes. Even when one of the input sources is unavailable, the converter topology allows for the powering of both loads. The proposed topology has fewer component counts and fewer voltage stresses than the standard topologies, which increases converter efficiency [49]. It can be used in step-up and step-down modes. In [50], the converter can be used in UPS applications since it is designed to be used in conjunction with H-bridge cells rather than other types of DIDO converters that are already in use. The capacity to serialize several voltage sources as the input helps the circuit's output voltages reach higher levels while also ensuring redundancy. However, in this design, adding more input voltage sources is simpler and requires fewer parts. Consequently, it is simpler to implement this topology.

The H-bridge cell2 in the DIDO buck-zeta topology is changed to a voltage source to increase the circuit efficiency. [51] introduces a straightforward converter capable of generating two distinct voltage levels from two different energy sources at the input side. It is crucial to optimize the size and cost of power electronic converters to guarantee the effective and dependable operation of electric vehicles (EVs) and drives. At least two separate DC voltage levels are required for EVs and some drive applications, one for the electric motor and the other for auxiliary loads. It can supply power to EV loads as well as perform on-board battery charging as explained in [52]. The number of input and output terminals can be selected as per the vehicle requirement.

A single-stage non-isolated bidirectional four-port buckboost converter with a smaller number of switches with integration of diversified sources in [53]. As illustrated in Fig. 6 (a), the SI-DIDO DC-DC converter is utilized to regulate the input voltage V_{PH} to the MPP voltage V_{PH} , and MPP and to produce a steady output voltage V_O to power the load. The converter will also manage and schedule the energy supplied by the two inputs piezoelectric transducer (PZT), and solar photovoltaic (PV) cells and the energy spent by the two outputs in [54]. A cutting-edge single-inductor dual-input dual-output (SI-DIDO) DC-DC converter for solar energy harvesting applications that lowers system costs and boosts overall power transfer effectiveness.

The single inductor is scheduled for the energy transfer among the dual inputs and dual outputs in each clock cycle rather than being shared in a multicycle-prescheduled timeslot way. As a result, each clock cycle regulates V_{PH} and V_O , and voltage ripples are minimal. This allows a significant reduction in the size of a capacitor connected to V_{PH} and V_O [55]. The battery provides one of the inputs, and its voltage is nearly constant. The lighting environment has an impact on the other PV cell input.

The converter VO-regulated output voltage ought to remain rather stable when the lighting changes. Once more, in this section, we solely consider line regulation that doesn't result in switching mode. Change the loading current and evaluate the converter's performance to address the line regulation that results in mode switching. The system is running with a lighting condition of 33700 LUX and a switching load current between 4 and 7.5 mA. The signal mode is changed to 1 by the mode detection circuit when it notices the change. The

TABLE 3.	Comparison	of various	topologies	based on	dual inpu	t single ou	tput converters.
----------	------------	------------	------------	----------	-----------	-------------	------------------

		Туре			Comp	Output	Effici	Valta	
Refer	Year	Input		Output	onent	Power/	ency	ge	Advantages
ence		1	2	1	s	Voltage	(%)	Gain	
[35]	2022	DC	DC	DC	16	200 W	95.87	High	Low voltage stress.High efficiency.
[36]	2022	PV	Grid	Battery	10	300 W	91.5	9.54	 Easy to design, Manipulation, and implementation process. Low switch voltage stress. Comparatively low cost.
[37]	2023	Battery	Superca pacitor	DC Link	13	5 kW	91.8	High	 Any port can transfer power. power flow is independently controlled. Size and cost are low.
[40]	2019	DC	DC	DC	16	293 V	85	High	 Continuous input current with bidirectional energy flow. Lower current stress and losses
[41]	2022	DC	DC	DC	8	500 W	-	High	 Continuous current for both inputs. Simple structure. High voltage gain.
[42]	2022	PV	Battery	DC	9	116mV	-	-	 Low cost and volume, portability of converter. Direct integration of sources is done, instead of cascaded topology. Complexity is less.
[43]	2021	DC	DC	DC	12	48 V	-	4	 Load current ripple is low. Isolates input sources. Gain boosting.
[45]	2020	PV	Battery	DC	13	200 W	93.6	2	 PWM Switching methods used. Control strategy is simple.
[46]	2021	DC	Battery	DC	14	100V	97.22	2-2.5	 Simple circuit. Low Volume, High efficiency. All switches operate under soft switching
[47]	2022	PV	Battery	AC	14	300~700 W	-	-	 Only differential power needs to be handled by dc – dc converter. High efficiency and integration, and lower cost can be achieved.
[48]	2022	DC	Battery	DC	17	535 W	94.88	High	 Soft switching performance is achieved. Voltage stress across the switches and diodes are low.

voltages of V_{PH} , $_{OP}$, and V_O experience re-regulation in the updated mode, indicating that the converter has switched to the heavy load mode of operation. The comparison of various dual input dual output converter topologies is discussed in Table 4.

C. MULTI INPUT SINGLE OUTPUT DC-DC CONVERTERS

The necessity and importance of a non-isolated converter don't have any galvanic isolation between the source and load sides, therefore the variations on the source side directly affect the load side of the converter. In contrast, isolated converters have a separate ground, which means that the input does not impact the output side of the converter. This enables them to operate effectively with a wide range of inputs and deliver a stable output. One common reason for using isolated converters is to ensure safety. By isolating the input and output ports, these converters enhance consumer safety on the load side and prevent short-circuit currents from flowing back to the source side. Isolated converters incorporate transformers into their design to achieve this isolation. Fig. 6 (b-c) shows the basic structures of the MISO converter. Fig. 6 (d) indicates the modular high voltage structure of a multi-input non-isolated converter. The comparison of various MISO converter topologies is discussed in Table 5.

A single-input single-output converter typically requires one transformer, while a MIMO converter may require multiple transformers [58]. However, this results in larger volumes, higher costs, lower power density, and reduced efficiency compared to non-isolated converters, which do not require such transformers. Another consideration is that isolated converters often require expensive measuring equipment, such as isolated probes, to measure currents. In the case of an isolated MIMO converter, multiple isolated probes are necessary, significantly raising the overall cost. Consequently, if consumer safety is not a concern, non-isolated converters prove to be a suitable choice for multi-input converters. Isolated converters also encounter various significant challenges, such as leakage inductance, core saturation, thermal effects, high voltage spikes across switches, and larger sizes, leading to higher costs compared to non-isolated converters. This study

			Т	уре		Compo	Output			
Refere	Year	Input		Output		nent	Power/	Efficien	Advantages	
nee		1	2	1	2	Counts	Voltage	cy (70)		
[49]	2018	DC	DC	DC	DC	11	80V & 60V	-	 Less number of components. Independent control of output voltage. 	
[50]	2021	DC	DC	DC	DC	13	40V & 5V	-	 Achieve higher level output voltages Implementation of circuit and increase the number of output voltage is easier. Less number of components. 	
[53]	2021	PV	Battery	Motor	DC	11	42.67V & 42.81V	98.2	 Less Component Count. Control strategy is simple and more reliable. Cost effective. 	
[54]	2019	PV	PZT	Battery	DC	15	180 mW	88.1	 Extra energy back to battery. Lower static power consumption 	
[56]	2021	DC	DC	DC	DC	11	90V & 75V	92.3	 Improved dynamic behavior and utilize maximum energy 	
[57]	2019	PV	Battery	Battery	DC	13	120 W & 20 W	-	 Cross regulation will be reduced by improving transient response. 	

TABLE 4. Comparison of various topologies based on dual input dual output converters.

TABLE 5. Comparison of various topologies based on multi input single output converters.

		Ту	pe	Compon	Output	Efficie				
Refer ence	Year	Input	Output	ent	Power/	ncy	f _s (kHz)	Voltag e Gain	Advantages	
ence		DC/ AC	DC/ AC	Counts	Voltage	(%)	(MIL)	e ouin		
[58]	2021	DC	DC	10N	450 W	93.5	30	15.66	 By increasing the number of inputs or the turns ratio of connected inductors, voltage gain can be enhanced. Voltage stress is low. 	
[59]	2023	DC	DC	8N+1	300 W	93.3	40	-	 Current stress on each input is low. Size of the converter is reduced. 	
[63]	2021	DC	DC	16	4 kW	96.1	20	-	 Input sources can be used individually or combine to enlarge total power. Total cost of energy is reduced. Prolonging of battery lifetime. 	
[64]	2021	DC	DC	7N	1.1 kW	91	15	-	Cost effective.High power and low power application.	
[65]	2020	DC	DC	9	400 W	96.37	100	-	 Simple structure. Component count of power conversion is reduced. High flexibility design. 	
[77]	2019	DC	DC	4N+6	200 W	98	120	8	 Less number of active switches. Light weight and less expensive. 	
[79]	2023	DC	DC	13	600 W	96.77	20	5	 Low voltage stress. Interface to allow any number of input sources. 	
[80]	2017	DC	DC	16	80 W	85	30	4	In the absence of one or two resources, the converter can supply the required power to the load.	

*N- Number of Modules

introduces a multi-input DC-DC converter designed for integrated PV applications, where the isolation of input and output terminals is not required.

The key attributes of the proposed converter in this paper are its high-power density, compact size, high efficiency, and affordability. In this scenario, a non-isolated topology is the optimal choice. Despite certain drawbacks like a high duty cycle ratio, low voltage gain, and additional circuitry in non-isolated DC-DC converter topologies, it is widely acknowledged in the literature that non-isolated topologies



FIGURE 6. (a) SI-DIDO DC – DC Converter. (b) Basic Structure of MISO Converter. (c) MISO DC – DC Converter. (d) Modular high voltage multi input converter [66].

are more advantageous and feasible for renewable energy applications compared to isolated converter topologies. The objective of this research is to address existing technical challenges and enhance the performance of non-isolated DC-DC converters in renewable energy applications. The MISO converter utilizes a single control system to achieve maximum power point tracking (MPPT) for PV sources, eliminating the need for multiple MPPT controllers as in a typical PV system. There are different converters available in the literature that take multiple inputs [58], [59], [60], [61], [62], [63], [64], [65], [66], [67], [68], [69], [70], [71], [72], [73], [74], [75], [76], [77], [78].

Due to the practical constraints of a normal boost converter's voltage gain and its high-power loss at severe duty cycles, significant research has been undertaken on high step-up DC-DC converters. During boost mode operation the input current flows through the inductor and to achieve a high voltage conversion ratio for diode-capacitor voltage multiplier cells (VMC) [59]. The converter [60] combines three-port boost and buck topologies to transmit power from and to energy storage units (ESUs) with continuous current, increasing the battery life as a single ESU. Additionally, switching capacitor technology and the primary boost switch are used in operating modes that provide power from energy storage units (ESUs) and energy generation units (EGUs) to the output along with increased voltage gain and continuous input current. A solar PV system with a fly-back converter and multiple ports. Because it includes many input sources and multiple output loads, the multiport system can typically transfer energy in both directions and improves the amount of possible energy utilization.

The fly-back converter [61] may convert between AC/DC and DC/DC, operating in both cases, and it appears to be flexible for any situation. The objective of PWM control approaches is to regulate the DC-bus voltage and achieve load-sharing control in alignment with the energy management strategy (EMS) [62]. However, the complex nature of these multi-input single-output converters presents challenges concerning reliability, efficiency, power density, control systems, cost, and device size [63]. The step-up n -n-stage DC-DC converters report Continuous current and almost high-voltage gain and can be expandable are the advantages of the converter proposed in [66] and [67], but it requires multiple input sources and many switches with the introduction of each stage making it unfit for various applications. Despite the high voltage gain of [68] and [70], it has considerably low normalised voltage stress (NVS) on its switches/diodes. Another important advantage of the proposed topology is that, as the number of input units increases, the voltage gain increases too, but the NVS on switches/diodes decreases. Less component count can reduce the size, mass and cost of the converter. Simultaneous or independent power transfer capability has been provided for input sources in [69]. Further, the bidirectional power flow capability has also been provided through the usage of a battery without any additional switches. A distinguishing characteristic of the DC-DC converter [71], [72], [73], [74], [75], [76], [77], [78], [79] is the utilization of an actively switched inductor-capacitor (ALC) network, a feature not commonly found in other multiport converters. However, multi-input DC-DC converters have

been extensively employed in various research areas concerning hybrid electric vehicles and fast charging of electric vehicles. These converters offer several advantages compared to conventional converters, including the reduction of power devices, the minimization of power conversion stages, and the centralization of control. A power electronics system is given for converting the electrical energy of a PV panel to a suitable form. A DC-to-DC converter architecture based on the integration of SEPIC and CUK converters has been presented, increasing the overall power output of renewable energy sources. If one of the input sources fails to operate, an effective energy management approach is applied to provide energy to the load. The converter has various advantages, including limited operation modes, no switching duty cycle limitations, and smaller component sizes. As an alternative to the centralized PV converter topology, the concept of PV distributed maximum power point tracking (DMPPT) has emerged to address power losses caused by irradiance mismatch and shading scenarios. In DMPPT technology, each module in a PV system is equipped with its converter and controller. This enables each module to independently perform maximum power point tracking (MPPT), maximizing the harvested power from the entire PV system. Parallel MICs with a single MPPT controller and current estimation have a greater component count, resulting in a bigger footprint, cost, and loss. The multi-input convert has the benefit of low part numbers and continuous input current, making it appropriate for renewable energy applications [76], nevertheless, it lacks bidirectional power flow.

The converter design presented in [78] is a multi-input single-output converter that addresses partial shading's drawbacks and enables the PV system to function at MPPT without the need for an extra energy buffer, such as a battery and it has a combination of series and parallel ones.

IV. MULTI-OUTPUT NON – ISOLATED DC-DC CONVERTER

The development of multiport converter topologies will be stated in this section. The converters have multiple output ports which have multiple loads. Furthermore, the multiport structures extended the opportunity of making the entire system simpler and more compact. This multi-output converter is further classified into SIDO, SIMO & MIMO.

A. SINGLE INPUT DUAL OUTPUT DC-DC CONVERTERS

In [81], a non-isolated SISO DC-DC converter is introduced, as shown in Fig. 7 (a). This converter has the unique capability to generate and regulate both boost and buck outputs simultaneously. An interesting observation is that the frequency of the ripple current flowing through the inductors is twice the switching frequency. Additionally, the voltages across all the switches are below half of the boost voltage and are found to be equal. This enables the utilization of high switching frequencies, resulting in a reduction in the size of passive components. Consequently, this converter exhibits advantages such as low switching stress, compact passive component sizes, and high efficiency. The primary benefits



FIGURE 7. (a) Single input dual output DC-DC converter. (b) Basic Structure of SIMO Converter. (c) SIMO configuration [87].

of the converter listed in Table 6 include the need for fewer components, the possibility to use a variety of sources with various voltage and current characteristics, and the capacity to obtain high voltage gain without using high-frequency switching. The comparison of various single input dual output converter topologies is presented in Table 7.

Lyapunov's strategy deals with analysing the behaviour of dynamical systems over time. With a focus on stability analysis, it helps us to understand how a system will behave under different conditions with input and outputs. In [82], mainly focus on voltage and current dynamics of boost and buck converter segments. The utilization of current-source-mode (CSM) converters to mitigate the issue of cross-regulation through topological modifications [83] has been increasingly prevalent in various domains, including light-emitting diode backlighting, wireless transceivers, hybrid power source systems resulting in severe performance degradation. The converter in [84] presents two proportionalintegral (PI) compensators applied to a dual output three-level converter (TLC), that combines the buck and boost converter, has low input current ripple and has critical issues, such

TABLE 6. SI-SIDO converters is derived from Buck, Boost, Buck-Boost.



TABLE 7. Comparison of various topologies based on single input dual output converters.

			Туре		Compo	Output	Effici				
Refer	Year	Input	Ou	tput	nent	Power/	ency	f _s (kHz)	Voltage Cain	Advantages	
ence		1	1	2	Counts	Voltage	(%)	(KI12)	Gain		
[81]	2020	DC	DC	DC	15	400 W	96.2	20	Medium	 Low Switching Stress. Input and output current ripples are low. Harmonic distortion is reduced by designing inductor filter. 	
[84]	2018	DC	DC	DC	15	300 W	95.9	20	Medium	 Voltage stress across semiconductor devices is low. Size of inductor shrank and volume of capacitors are small. ON-state resistance and diode reverse recovery losses are less. 	
[85]	2021	DC	DC	DC	12	150 W	91	50	High	 Lower output capacitor stress due to dual output. Input current that is constant. Switches and diodes are subjected to low voltage stress. 	
[86]	2020	DC	DC	AC	14	60 W	-	10	Medium	 Less number of semiconductor devices. Better shoot through protection compared to VSI. Continuous supply for both AC and DC outputs from a single DC source. 	
[89]	2018	DC	DC	DC	7	10 W	95.8	10	Medium	 Efficient and economical solution for volume of DC grid. Low voltage distribution and multiple voltage levels. 	
[92]	2023	DC	DC	DC	10	17 W	92.03	25	-	Faster load transient.Low cross regulation.	
[93]	2022	DC	DC	DC	11	120 W	-	20	-	 DVMC is effectively eliminates the CC problems. High power density and size is compact. 	

as very slow dynamic and large undershoots/ overshoots. The hybrid converter [85] is implemented by replacing a voltage source inverter (VSI) bridge network for the switch of a typical boost converter. Extensive research has been conducted in the literature to examine the impact of a coupled inductor on the performance of different converters, including high step-up, interleaved, and multiple-output converters. The investigation of inductor coupling's influence on the continuous conduction mode (CCM) and discontinuous conduction mode (DCM) boundary of the CI-SIDO boost converter is explored in [86], [87], [88], [89], [90], and [91]. The uni-input Duple-Output Buck Converter plays a major role in maintaining the voltage levels of the microgrid. The expressions of the average value of inductor currents at the boundary condition for Dt = 0 are presented in [91]. In certain applications [88] where a high AC voltage is required, the voltage-source inverter (VSI) can only function as a buck inverter because an additional DC-DC boosting stage is necessary to achieve high gain. However, the VSI may not be suitable for all applications due to limitations in available DC input voltage. Pseudo-continuous conduction mode single-inductor dual-output (SIDO) DC-DC converters often suffer from slow load transient performance, potential crossregulation, and low efficiency. To address these limitations, a voltage-ripple (VR)-based dynamic freewheeling (DF) control mechanism is proposed in [92] for hybrid conduction mode (HCM) SIDO buck converters. This mechanism aims to overcome the drawbacks associated with SIDO converters by improving their efficiency and load transient performance. Which can be generated by using a proportional-integral (PI) controller as a small-signal modelling-based cross-derivative state feedback controller proposed in [93].

B. SINGLE INPUT MULTI OUTPUT DC-DC CONVERTERS

DC-DC converters have recently been the subject of intensive research to increase their flexibility and dependability while at the same time lowering their overall cost. Incorporating both positive and negative layers, these converters can independently adjust voltage and achieve substantial amplification. Three different types of multiport DC/DC converters can be distinguished: those with numerous inputs and a single output, those with numerous inputs and multiple outputs, and those that combine both layouts [94], [95], [96], [97], [98], [99], [100], [101], [102], [103], [104], [105], [106], [107]. Multi-input multi-output DC/DC converters can use a variety of DC voltages with discrete V-I characteristics to provide a wide range of DC voltage levels at the output. The converter [99] uses a two-stage structure and a series connection of current-source-mode (CSM) converters to ensure strong individual control capabilities for a range of output voltage levels. A voltage-source-mode (VSM) buck converter is used in the first stage to reduce the voltage and deliver a steadily controlled, stable current. The second stage, in contrast, uses CSM converters that are coupled in series and have intrinsic independence from one another. By doing this, they can accomplish voltage step-up with a less complex control system and avoid any cross-regulation problems. The converter is often very simple to build, centrally regulated, incredibly reliable, and relatively affordable, and output ripple is less but synchronization is difficult. Non-isolated SIMO converters have the advantage of employing fewer components and switches to achieve a more compact circuit due to the need

for a transformer to provide electrical isolation. [95] presents an application to a type of non-isolated SIMO in which the averaged model of the converter is generated utilizing the notion of quasi-Weierstrass transformation in conjunction with consistency projection operators.

The varied voltage levels were acquired to power various DC loads such as mobiles, routers, nano stations, laptop chargers, and so on [97]. The fundamental disadvantages of multiport and multi-input conversion systems are that they convert several inputs to a single output, increasing the cost of renewable energy generation and power loss. Reference [98] suggested a new single-input and multi-output (SIAMO) model for renewable energy converters. Fig. 7 (b) shows the basic structure of a single-input multi-output converter. The comparison of various single input multi output converter topologies is discussed in Table 8.

The large difference in switching frequency between the input and output stages in the DF-SIMO topology makes the dynamic and steady-state behaviour of these output controllers and the expressions that govern their operation and performance fundamentally different than when these same controllers are used in single-frequency SIMO topologies [107]. Moreover, the design trade-offs between performance and circuit complexity become quite different in the DF-SIMO topology due to the high switching frequency of the output stage.

The optimum strategy for controlling such a power converter topology is subject to many considerations, such as circuit complexity, stability, expected variability in operating conditions, and desired steady-state and transient performance. The advantage of superior dynamic response compared to ripple-based charge-mode and average-based voltage-mode controllers. power- the stage is conceived and switched appropriately to generate the unique floating output among others. Regulated floating dual-slope (RFDS) drivers are proposed [108] to limit the di/dt through the switches and to reduce the switching noise.

Portable electronic devices require multiple supply voltages with relatively large differences in load currents, causing serious regulation, EMI, and efficiency problems when adopting a single-inductor multiple-output (SIMO) DC-DC converter. To resolve these issues, [109] presents a reconfigurable SIMO system that utilizes a dual-bus multipleoutput (DBMO) with ten-output having two buses, one for heavy-load and one for light-load outputs. By calculating the total energy needed by all outputs to draw the same amount of energy from the input, the output-voltage-aware charge control (OVACC) technique lessens cross-regulation. Many methods have been suggested to suppress cross-regulation, however, it is still unclear where the limit lies.

The differential evolution (DE) approach is then used in [110] and [111] to tackle this issue. Most SIMO converters face limitations concerning operating constraints related to duty ratio and inductor charging when producing multiple outputs. One particular challenge in the design of SIMO converters is the presence of cross-regulation issues.

TABLE 8.	Comparison	of various	topologies	based on s	single input	multi output	converters.
----------	------------	------------	------------	------------	--------------	--------------	-------------

		Т	уре	G	0.1.1	E CC		
Refer	Year	Input	Output	Compo nent	Output Power/	Efficie ncy	f_s	Advantages
ence		DC/ AC	DC/ AC	Counts	Voltage	(%)	(kHZ)	9
[94]	2022	DC	DC	4N	100 W	85.41	10	 Multiple outputs can be extended. At load variation, voltage regulation is better. Isolation of load
[96]	2022	DC	DC	4N	100 V	-	10	 Reliability and flexibility. Easy to implementation. Maximize the efficiency reduce the size and expense.
[97]	2022	DC	DC	3N	60 W	95	10	 It has better control methods and low cost. More efficient. It is flexible and reliable.
[100]	2021	DC	DC	3N+1	-	High	10	 Switching Loss and Cost is low. Reliability and availability are high.
[101]	2020	DC	DC, AC	5N+1	83.5	87.5	500	 Low cross regulation. Low overshoot and undershoot. Multi output hybrid power stage.
[102]	2018	DC	DC, AC	21	153V, 58V	-	5	 Fewer switches and passive components. Size is small. Low cost. Both DC and AC loads are used as outputs.
[106]	2022	DC	DC, AC	3N	20 W	88	400	 Simple, compact and low cost Less component counts. High power density. Better flexibility, scalability, and combinations of AC and DC loads.
[109]	2019	DC	DC	3N+1	2.83	88	1MHz	 Switching loss is low. In dual-bus multiple-outputs can be controlled.
[112]	2021	DC	DC	4N	100 W	-	5	 Low cost. Low voltage input power source. Different output voltages for low and high-power applications
[115]	2022	DC	DC	4N	200 W	-	50	 No constraints on inductor currents. Buck boost positive output voltage. Can generate three different output voltages. Simple structure.

However, this study introduces a novel SIMO topology aimed at addressing the aforementioned limitations as shown in Fig. 7 (c). In [113], [114], and [115], this topology offers the capability to generate three distinct output voltages without any constraints on the duty cycle or inductor currents. Importantly, this design effectively eliminates cross-regulation problems, ensuring that sudden fluctuations in the inductor and load currents have no impact on the output voltages. While Single Input Multi Output (SIMO) DC-DC converters offer several advantages, they also present some limitations or demerits: Limited Input Voltage Range, Complexity in Design, Cross-Regulation Effects, Potential Efficiency Trade-offs, Increased Component Count and Sensitivity to Load Variations. transitioning from SIMO to MIMO converters represents a significant advancement in power electronics technology, enabling more resilient, efficient, and flexible energy management solutions across a wide range of applications.

C. MULTI INPUT MULTI OUTPUT DC-DC CONVERTERS

A multi-input, multi-output DC-DC converter is a versatile power electronics device capable of handling power from multiple sources and providing regulated voltages to several components simultaneously. It excels in complex systems with diverse power requirements, integrating various energy sources efficiently. The converter focuses on high efficiency, independent control for each output, and adaptability to different industries. Its compact design, reliability features, and advanced control algorithms make it suitable for applications in telecommunications, automotive, renewable energy, and aerospace, offering customization options to meet specific needs. MIMO provides power supply configuration versatility, allowing for a range of input and output combinations to be used in complex applications. Suitable for incorporation into intricate systems where various parts or subsystems have distinct power needs. Despite managing several inputs and outputs, aims for a small form factor to satisfy space requirements in a range of applications. For the design of multiport converters in electric vehicles (EVs) and portable electronic applications, these integrated models appear to be quite attractive. However, it has assumptions on (i) output current $(i_{01} > i_{02})$ and (ii) due to the limitation of output current restricted the output voltage (i.e. $V_{i1} > V_{o2}$), other the inductor is continuously charged. Due to the intrinsic working principle and lack of restrictions imposed by traditional twoport converters, the investigation of numerous conceivable

SI-MIMO topologies is possible. The automated process's use of computer codes makes the topologies' rigorous derivation possible [116], [117], [118].

In parallel, 3D images were captured utilizing two distinct channels with overlapping field-of-views (FOVs). Using this method, [119] was able to simultaneously capture photos with big FOV and low resolution and small FOV and high quality. Developed a general unique model for the MIMO hybrid energy system, decoupling network, sub synchronous oscillations (SSOs) and systematic design of the linear controllers are introduced in [123] and [124]. The MIMO converters using Augmented Sinusoidal Pulse Width Modulation (ASPWM) technique, constraints on structure of state space controller matrices, relative gain array (RGA), participation matric (PM) method, Hankel interaction index array (HIIA), Linear matrix inequality (LMI), MPPT and PSO optimization algorithms to determining the phase sequence and integrated in working of DC-DC converters [128], [129], [130]. The dependability and stability issues related to the power generated by renewable energy conversion systems have been effectively handled and stabilized using a power converter system setup. Furthermore, the intermittency problems that could occur during the generation process have been successfully overcome using multiport converters [135], [136].

Fig 8 (a) shows the basic structure of a multi-input multioutput converter. A MIMO converter with a simple structure (Fig. 8 (c)) is described in [125], [132], and [138]. This setup successfully integrates different voltage and current characteristics of two energy sources using a time multiplexing concept. However, it has a drawback, requiring n+m+1switch for n-inputs and m-outputs, with outputs connected in series. Additionally, during CCM operation, it faces crossregulation issues. A Model Predictive Control-based MIMO converter is suggested in [139] as a solution to the issues raised in [133] and [137]. With the use of this control strategy, cross-regulation issues, voltage ripples, and converter output voltage regulation are all avoided. The comparison of various multi-input multi-output converter topologies is discussed in Table 9. A novel MIMO topology designed for electric vehicle applications is introduced in [140] and [141]. This configuration ensures that both output voltages exceed the input voltage, with the output switches restricting their levels. However, a limitation of this setup is that only one input source can supply energy at any given moment, potentially restricting the possibility of wide-ranging duty cycle operations. Fig. 8 (b) for a representation of the schematic circuit of the MIMO converter presented in [146].

 G_{max} is the highest value in one port of voltage gain is higher than that of other ports in the converter. The normalized maximum voltage stress on the switch is $V_{S(max)}$, $I_{S(max)}$ is the average current stress on the switch and n_{out} is the number of output ports. The comparison of various high voltage gain with the advantages of DC-DC converter topologies is discussed in Table 10.

A different single-stage MIMO converter with a robust control approach is provided in [147] and [131] that uses



FIGURE 8. (a) Basic structure of MIMO converter. (b) Schematic circuit of MIMO converter. (c) MIMO configuration.

only one inductor, smaller in size and less expensive as a result. However, it requires an n+m+1 switch for n inputs and m outputs, which poses efficiency problems due to large switching losses and raises the converter's total complexity.

TABLE 9.	Comparison o	f various topologies	based on multi input	multi output converters.
----------	--------------	----------------------	----------------------	--------------------------

Refer ence	Year	No. of Input	No. of Output	Compon ent Counts	Output Power/ Voltage	Efficie ncy (%)	f _s (kHz)	Advantages
[117]	2022	N-1	N	12+6N	225W, 250 W, 410 W	-	50	 High gain in voltage. Switch stress is minimal.
[121]	2021	N	N	7N	-	-	50	 Voltage and current stress on OFC are low. Negative voltages can supply. Simplifying in design and saving cost.
[122]	2020	N	N	12N	500 W	95	200	 Multicell reconfigurable maintaining high performance. Low power conversion stress. Matrix method reduced the circuit complexity.
[125]	2020	N	N	6N+1	2.4 W, 28 W	88	100	 Simple control, high flexibility No cross regulation. Extension capability is high. Low complexity control. Medium power density.
[126]	2019	N-1	N	10N	1 kW	96.8	40	 Switched capacitor to generate different output voltage level. Operate at low/ high power ranges. Low voltage stress across diodes and switches.
[127]	2018	N	N	7N	250 W	93.5	250	 Systematic synthesis, point to point power routing. Common ground between all ports. Less element counts.
[129]	2022	N	N	11N	800 W, 50W	-	40	Large AC and DC output voltages can be obtained by cascading several micro-hybrid converters.
[134]	2021	N	N	8N+4	24 V, 12 V, 5 V	-	40	 Buck boost converter with any number of inputs and outputs.

TABLE 10. Comparison of high voltage gain converters.

DC-DC Converte rs	G _{max}	I _{S(max)}	V _{S(max)}	n _{out}	Duty cycle operation	Controllable output voltages at the same time	Expandable from input and output sides
[117]	$\frac{1}{(1-D)^2}$	$\frac{D_5 I_{05}}{1 - D_5}$	V ₀₅	5	$0 < D_1, D_2, \\ D_3, D_4, \\ D_5 < 1$	Yes	N-1 input and N output
[125]	$\frac{D}{1-D}$	$\frac{I_0}{1-D_4}$	<i>V</i> ₀₁	4	$0 < D_1, D_2, D_3, D_4 < 1$	Yes	N input and N output
[126]	$\frac{4}{1-D}$	$\frac{I_{01} + I_{02}}{1 - D_4}$	$\frac{V_{01}}{4}$	2	$0.5 < D_1, D_2, D_3, D_4 < 1$	Yes	N input and M output M <n< td=""></n<>

 $*G_{max}$ – Total voltage gain, $I_{S(max)}$ – Current stress on switch, $V_{S(max)}$ – maximum voltage stress on switch, n_{out} – number of output ports.

The synthesis of MIMO converters involves a three-step process: (i) Selecting and connecting the PSCs (Power Semiconductor Components). (ii) Choosing a configuration for the OFC module (Output Filter Components). (iii) Connecting the PSC and OFC modules to create the MIMO converter [120], [121], [122]. Moreover, there is a multicell reconfigurable multi-input multi-output (MR-MIMO) power conversion architecture designed for various multiport applications, including multisource energy routers, battery balancers, and photovoltaic optimizers. In [137], a matrix converter-based MIMO buck-boost converter is introduced for microgrid applications. The original design is later expanded to support n-inputs and n-outputs after being first designed for a certain number of inputs and outputs. Unfortunately, this addition increases the size of the inductor and the number of switches, creating a complex system with increased costs and losses. The converter operates on the time-sharing concept, which may limit the efficient utilization of energy sources. In response to these challenges, an alternative diode-clamped converter configuration-based MIMO converter is presented in [142]. The outputs in this particular structure are adaptable and independent in their functionality. This design's disadvantage is that it calls for the employment of extra switches, which leads to higher losses, a more complex structure, and a possible sensitivity to cross-regulation problems during CCM operation. To integrate multiple energy sources, achieve controlled output, and support bidirectional flow, an alternative MIMO converter is suggested in [143]. However, a drawback of this converter is that to avoid incorrect operation, both energy sources must run within specific duty cycle restrictions.

A MIMO boost converter designed primarily for applications involving renewable energy sources is introduced in [144]. However, it mandates the usage of n+m+2 switches and necessitates additional energy storage components when set for n-inputs and m-outputs. The calculation of these parameters in different types of non-isolated converters along with output voltage and voltage gain were presented in Table 11.

The converter grows and complexity as a result. A different strategy is provided in [145], where a converter is created by combining the traditional converter synthesis of DC-DC converters with dual input and single output (DISO) and single input and multiple output (SIMO). Several techniques have been developed to increase voltage levels, resulting in the creation of different types of high step-up DC-DC converters. There are a wide variety of bidirectional converters, some of which are classified and evaluated in [148]. In [149] enables the incorporation of various renewable energy sources, each differing in type and capacity, into a bipolar medium voltage DC microgrid. Its primary benefits lie in its high-power density and the diminished quantity of switches required compared to the distinct converters. To improve the voltage gain of power converters with proper duty ratio, many techniques, such as coupled inductors (CIs), diode-capacitor voltage multipliers (DCVMs), and their combinations, have been investigated. Using these techniques not only improves the voltage gain but also reduces the voltage stress across semiconductor elements. On the contrary, Multiphase interleaved high step-up DC-DC converters are widely studied in [150], which have better efficiency characteristics under heavy load. Indeed, the investigation of various multiport DC-DC converter configurations reveals that there is no universal structure that fulfils all the objectives, such as costeffectiveness, reliability, part count reduction, flexibility, and modularity, simultaneously.

Each topology has its own set of advantages and disadvantages, making them suitable for specific applications and scenarios. The precise needs and limitations of the intended application determine the choice of a particular multiport DC-DC converter design. To choose the topology that is most appropriate for their specific use case, engineers and researchers must carefully analyse the benefits and drawbacks of each topology. The required features and trade-offs must be balanced while considering aspects like cost, dependability, part count, flexibility, and modularity. The performance of the converters can be defined in terms of efficiency, switching losses and also the voltage stress on components. These parameters differ concerning the number of active and passive elements used, types of switches, switching frequency, inductor, capacitor design, and so on. Along with this, the converter functions like a buck, boost, and buck-boost rely on the capacitor and inductor design. These factors involve defining the boosting factor of the converter. Different non-isolated converters were discussed in the literature and their corresponding design considerations are given in Table 12. The non-isolated dc-to-dc converter is compared with features, benefits, and drawbacks of different topologies are discussed in Table 13.

V. CONTROL TECHNIQUES

Control techniques play a crucial role in achieving maximum efficiency for non-isolated DC-DC converters. These techniques optimize the overall operation of the converter topologies. The control parameters for DC-DC converters include input voltage, duty cycle ratio, reference voltage, and output voltage. Based on these parameters, a specific control technique can regulate the operation of the converters. For low input and output voltages, the control technique increases the duty cycle ratio to perform voltage step-up operations according to the reference output voltage. Conversely, if the input and output voltages are higher, the control technique reduces the duty cycle to perform voltage step-down operations. The control technique is applied in various scenarios to achieve optimum control of the DC-DC converters, ensuring that the switching operation meets the output requirements [151]. Numerous control techniques with different features, such as response time, efficiency, and robustness, are available [152], [153]. Different converter topologies possess unique characteristics that are considered before integrating them with DC-DC converters. However, it is not possible to optimize all aspects of the control technique simultaneously, necessitating trade-offs for certain parameters based on the priority of application requirements. Furthermore, specific converter topologies are best suited with control techniques [190], making them a significant combination for various applications.

A. PULSE WIDTH MODULATION CONTROL

Pulse width modulation (PWM) has been one of the most conventional modulation techniques for switching DC-DC converters. It compares the error signal with a sawtooth wave to generate the control pulse. Conventional PWM control technique suffers from slow dynamic response due to smooth error signal variation under step line or load variation. The pulse width modulation DC-DC converter attains a significant increase in voltage output using the right duty ratio, along with a coupled inductor and voltage multiplier technique.

The energy stored in the leakage inductor of the coupled inductor can be reused. Additionally, as both the primary and secondary switches can be activated with zero-voltage

Efficiency	Voltage Stress/ Switching Loss	Output voltage	Voltage Gain	Refer ences
$\frac{P_o}{P_o + P_{SW} + P_D + P_L + P_C}$	$ \left(\frac{f_{S}(t_{ON} + t_{OFF})}{2D(1 - D)} + \frac{f_{S}(t_{ON} + t_{OFF})}{2(1 - D)} + \frac{R_{S1}D}{(1 - D)^4R_o} + \frac{R_{S2}D}{(1 - D)^2R_o} \right) P_o $	$\frac{V_{C1}D}{(1-D)}$	$\frac{D_1}{(1 - D_1)(1 - D_i)}$	[5]
$\frac{P_o}{P_o + P_{loss}}$	$\frac{(1+D)V_o}{3+D}$	$\frac{3+D}{(1-D)^2}V_{in}$	$\frac{3+D}{(1-D)^2}$	[4]
$\frac{\frac{V_2/V_1}{V_1}}{\frac{(1+d_1)}{(1-d_1-d_2)} + \frac{2\pi V_1^2 C_1 d_1}{T_s} + P_{SW}}$	$\frac{1}{T_{S}} \begin{pmatrix} (t_{r1} + t_{f1})(V_{s1} * I_{S1}) \\ + (t_{r1} + t_{f1})(V_{s1} * I_{S1}) \\ + (t_{r1} + t_{f1})(V_{s1} * I_{S1}) \end{pmatrix}$	$\frac{V_1(1+d_1)}{R(1-d_1-d_2)} + V_1 I_{C1} d_1 + P_{S-SW}$	$\frac{\frac{(1+d_1)}{(1-d_1-d_2)} + \frac{2\pi V_1^2 C_1 d_1}{T_s}$	[23]
$\frac{M(1-D)}{2(n+1)}$	$\frac{V_0}{2(n+1)}$	2n + 2	$\frac{2(n+1)}{1-D}$	[28]
$\frac{1}{1 + \binom{P_{loss}}{P_o}}$	$\frac{V_0}{M+1}$	<i>M</i> + 1	$\frac{M+1}{(1-D)^2}$	[29]
$\frac{P_o}{P_o + P_{SW} + P_D + P_L + P_C}$	$\frac{1}{2}I_{rms}^2 R_{Ds,on} + \frac{1}{2}C_{oss}V_{DS}^2 f_{sw} + C_{snub}V_{DS}^2 f_{sw}$	$\frac{1}{2} \left[\frac{V_1}{2} \left(n + \sqrt{n^2 + \frac{4D_1^2}{k_1}} \right) + \frac{V_2}{2} \left(n + \sqrt{n^2 + \frac{4D_2^2}{k_2}} \right) \right]$	$\frac{n(D_1+D_1')}{D_1'}$	[34]
$\frac{P_o}{P_o + P_{loss}}$	$\frac{k}{5Nk + (7k - 2)}V_0$	$\frac{5Nk + (7k - 2)}{1 - D}V_{in}$	$\frac{5N+5)}{1-D}$	[35]
$\frac{P_o}{P_o + P_{loss}}$	$f_s C_s \left(\frac{V_{in}}{(1-d)}\right)^2$	$\frac{(1-d_1)V_{in1}}{(1-2d_1)}, \frac{(1-d_1)V_g}{(1-2d_1)}$	$\frac{(1-d)}{(1-2d)}$	[36]
$\frac{P_o}{P_o + P_{loss}}$	$\frac{1}{(1-D)G_t} + \frac{1+n_s D}{(1-D)^2 G_t}$	$\frac{\frac{(1+n_{s1})(1+n_{s2})D}{(1-D)^2}V_{i1}}{+\frac{1+n_{s2}D}{1-D}V_{i2}}$	$\frac{(1+n_s)^2 D}{(1-D)^2} + \frac{1+n_s D}{1-D}$	[58]
$\frac{V_o I_o}{V_o I_o + (I_{La}^2 + I_{Lb}^2)R_L + (I_{Sa}^2 + I_{Sb}^2 + I_{Sc}^2 + I_{Sd}^2)R_{ON} + (I_{Da}^2 + I_{Db}^2 + I_{Dc}^2 + I_{Dd}^2)R_D}$	$ \begin{array}{c} V_{0}(1-\alpha_{4})+V_{1}(3\alpha_{2}-1)+V_{2}(2\\ -\alpha_{1}+3\alpha_{2})\\ +V_{3}(1\\ -2\alpha_{3}) \end{array} $	$\frac{[1 + (\alpha_2 + \alpha_3)(\alpha_4 - 2)]V_1 + (2 - \alpha_4)[(\alpha_1 + \alpha_2)V_2 + \alpha_3V_3]}{(1 - \alpha_4)^2}$	$\frac{D}{2(1-D)}$	[79]
$\frac{P_u}{P_a}$	$R_{on_{Mos}}I^2_{L_{ieff},(0-d_2T)}$	$\begin{split} \widehat{V_1} &= H_{11}\widehat{d_1} + H_{12}\widehat{d_2} + P_{11}\widehat{u_e} \\ &+ P_{12}\widehat{\iota_{ch1}} + P_{13}\widehat{\iota_{ch2}} \\ \widehat{V_2} &= H_{21}\overline{d_1} + H_{22}\overline{d_2} + P_{21}\widehat{u_e} \\ &+ P_{22}\widehat{\iota_{ch1}} + P_{23}\widehat{\iota_{ch2}} \end{split}$		[84]
$\frac{P_O}{P_O + P_S + P_C}$	$\frac{1}{6}R_L f(t_{on} + t_{off})P_0$	$V_{01} = D_1 V_{DC}$ $V_{02} = D_2 V_{DC}$		[94]
	$\frac{V_o}{m}$	$\frac{m}{1 - D_{n+1}} [D_1 V_1 + \sum_{j=2}^n (D_j - D_{j-1}) V_j]$	$\frac{m}{1-D_{n+1}}$	[142]

TABLE 11. Parameter comparison of non-isolated DC-DC converter topologies.

switching, the soft-switching technique helps in decreasing switching losses [178]. A simple Pulse Width Modulation (PWM) control technique is used in DC-DC converters to

efficiently regulate the output voltage. It involves varying the duty cycle of a high-frequency square wave generated by a switching element (such as a MOSFET) in response to

TABLE 12. Design consideration of non-isolated DC-DC converter topologies.

Inductor Design	Capacitor Design	Refe rence
$L_{m1} \ge \frac{D(1-D)V_{in1}}{(5N+5)I_o f_s},$ $L_{m2} \ge \frac{D(1-D)V_{in2}}{(5N+5)I_o f_s}$	$C_{1} \geq \frac{NI_{o}D}{1\%(1-D)V_{c1}f_{s}} , \qquad C_{2} \geq \frac{2I_{o}D}{1\%(1-D)V_{c2}f_{s}}$ $C_{3} \geq \frac{3I_{o}D}{1\%(1-D)V_{c3}f_{s}} , \qquad C_{4,5} \geq \frac{I_{o}D}{1\%(1-D)V_{c4,5}f_{s}}$ $C_{o} \geq \frac{I_{o}D^{2}}{2\%(1-D)V_{c0}f_{s}}$	[35]
$L_{1g} = \frac{R(1-2d)}{x_{L_1}\% f_s d}, \qquad L_2 = L_3 = \frac{R(1-2d)}{x_{L_{2,3}}\% f_s d}$	$C_1 = \frac{d}{x_{c1}\%R(1-2d)f_sD} , \qquad C_2 = C_3 = \frac{d}{x_{c1}\%R(1-2d)f_sD}$	[36]
$L_{1a} = \frac{(1-d_1)^2 d_1 V_1 R}{2V_0 f} , \qquad L_{1b} = \frac{(2-d_1) d_1 V_1 R}{2V_0 f}$ $L_{2a} = \frac{(1-d_2)^2 d_2 V_2 R}{2V_0 f} , \qquad L_{2b} = \frac{d_2 V_2 R}{2V_0 f}$	$C_{1} = \frac{V_{o}}{RfV_{1}(x_{C_{1}}^{\%})}, \qquad C_{2} = \frac{d_{2}V_{o}}{RfV_{2}(x_{C_{2}}^{\%})}$ $C_{m1} = \frac{V_{o}}{Rf(x_{C_{m1}}^{\%} * v_{C_{m1}})}, \qquad C_{o} = \frac{d_{1}}{Rf(x_{C_{o}}^{\%})}$	[40]
$L_{1} = \frac{V_{PV}}{\Delta i_{L1} f_{s}} k_{S1} , \qquad L_{2} = \frac{V_{PV} + V_{C1}}{\Delta i_{L2} f_{s}} k_{S2}$	$C_{1} = \frac{V_{o}}{R\Delta V_{C1}f_{s}(1-k_{s1})}k_{s1} , \qquad C_{o} = \frac{V_{o}}{R\Delta V_{Co}f_{s}}k_{s2}$	[45]
$L_{a} = \frac{V_{2}(\alpha_{1} + \alpha_{2}) + V_{3} \alpha_{3} + V_{1}(\alpha_{4} - \alpha_{3} - \alpha_{2})}{f \Delta i_{La}}$ $L_{b} = \frac{V_{2}(\alpha_{1} + \alpha_{2}) + V_{3} \alpha_{3} + V_{1}(\alpha_{4} - \alpha_{3} - \alpha_{2})}{f \Delta i_{La}}$		[79]
$L_{1} = \frac{TV_{in}(D_{2} - \frac{1}{2})}{\Delta i_{L1}}$ $L_{2} = \left(\frac{V_{01}}{2} - V_{02}\right)\frac{T(1 - D_{1})}{\Delta i_{L2}}$	$C_{1} = \left(i_{L1} - i_{L2} - \frac{V_{01}}{R_{01}}\right) \left(\frac{1 - D_{1}}{\Delta V}\right) T$ $C_{2} = \left(i_{L1} - \frac{V_{01}}{R_{01}}\right) \left(\frac{1 - D_{1}}{\Delta V}\right) T$	[81]
$L_1 = L_2 \ge \frac{RD(1-D)^2}{16f_s}$	$C_1 = \frac{2V_0}{R\Delta V_{c1}f_s} , \qquad C_2 = \frac{DV_0}{R\Delta V_{c2}f_s}$ $C_3 = \frac{DV_0}{R\Delta V_{c3}f_s} , \qquad C_4 = \frac{(1+D)V_0}{R\Delta V_{c4}f_s}$	[85]
$L_{1min} = \frac{R_{L1max}(1 - D_{1min})}{2f}$ $L_{2min} = \frac{R_{L2max}(1 - D_{2min})}{2f}$	$C_{1min} = C_{2min} = \frac{D_{max}}{2r_c f_s}$	[94]
$L_1 = \frac{V_i D_1}{f \Delta I_{i,1}}$ $L_2 = \frac{V_i D_2}{2f \Delta I_{L2}}$	$C_1 = \frac{V_1 D_1}{f R_1 \Delta V_1}$ $C_2 = \frac{V_2 D_2}{2f R_2 \Delta V_2}$	[112]
$L_{1min} = L_{2min} = \frac{2}{27} \frac{R_{Lmax}}{f_s}$ $L_{3min} = \frac{R_{Lmax}(1 - D_{min})}{f_s}$	$C_{1min} = \frac{D_{max}V_{01}}{V_{CPP}R_{L1max}f_s} , \qquad C_{2min} = \frac{D_{max}V_{02}}{V_{CPP}R_{L2max}f_s}$ $C_{3min} = \frac{D_{max}}{2r_cf_s}$	[115]
$L_{1} > (V_{i}D_{1})/(2I_{L1}f_{s})$ $L_{2} > (V_{i} - V_{02})(1 - D_{2})/(2I_{L2}f_{s})$	$C_{1min} = \frac{D_1 T_s (l_{01}/D_1)}{\{0.01V_{C_1} - r_{C_1}l_{01}/[D_1(1 - D_1)]\}f_s}$ $C_{1min} = \frac{[I_{L1} - I_{02}/(1 - D_1)](1 - D_1)}{\{0.01V_{C_2} - r_C[I_{L1} - I_{02}/(1 - D_1)]/D_1\}f_s}$ $C_{01ESR} = \frac{(1 - D_1)I_{01}}{[(0.01V_{C01} - r_c I_{01}/D_1)f_s]}, C_{01THT} = \frac{1}{[0.01R_{01}(0.1f_s)]}$ $C_{02ESR} = \frac{(D_2 I_{02})}{[(0.01V_{C02} - r_c I_{L2})f_s]}, C_{02THT} = \frac{1}{[0.01R_{02}(0.1f_s)]}$	[117]
$L \ge \frac{V_{01} + V_{02}}{2I_L f_s}$	$C_1 > \frac{I_{01}D_{S1}}{2V_{01}f_{S1}}$, $C_2 > \frac{I_{02}D_{S2}}{2V_{02}f_{S2}}$	[125]
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$C_{out_k} = \frac{I_{out_k}}{f_s \Delta V_{out_k}}$	[126]

TABLE 13. Comparison of non-isolated DC-DC converter topologies.

Converter Topology	Features	Benefits	Drawbacks	Refere nces
High Gain Transformerless Double-Duty-Triple- Mode DC-DC Converter	Complexity is Medium It Requires Precise Control. Cost is Moderate. Size is Medium.	Output is non-inverting. High output gain without the use of sophisticated procedures. applications with low input power (RES)	Voltage fluctuations during switching modes. Transient problem is brought on by capacitor charging and diode reverse recovery. Control the double duty cycle requires efficient control technique. Unidirectional power flow.	[23]
The High Gain Three- State Switching Hybrid Boost Converter	Complexity is Medium It Requires Precise Control. Cost is Moderate. Size is Medium.	Output is non-inverting. Duty cycle ratio is low. Output gain is high with voltage lift technique. Utilized for renewable energy application.	Control the double duty cycle requires efficient control technique. Unidirectional power flow. Transient problem is brought on by capacitor charging and diode reverse recovery. Transitions in the duty cycle are changed.	[24]
High step-Up DC-DC Converter	Complexity is Medium It Requires Precise Control. Cost is Moderate. Size is Medium.	Output is non-inverting. Capacitive filter values required less. Low input ripples and soft switching. Common ground for input and output.	Suitable for multi-input configuration. None of the switches should have a duty cycle of less than 50%. Due to coupled inductors, input conduction losses will occur.	[28]
Non-isolated high gain Quadratic boost DC-DC converter	Utilizes one switch that is active. High conversion ratio for voltage. Low voltage puts semiconductors under stress.	Diode -capacitor multiplier (VM cell) is added. Simplifies gate drive circuit, since additional switch is not added. Depends on number of multiplier cells.	Large voltage gain can achieve with coupled inductor.	[29]
Dual input DC-DC Converter	Medium complexity. Elimination of magnetic inductance saturation.	Soft-switched and integrate two low voltage energy sources. Voltage fed input structures with high current.	Dynamic performance load variation and source disturbance. Inherent overshoot, steady state error Suitable for multi-input configuration.	[34]
Couple Inductor High Step-Up DC-DC Converter	Independent power transfer to load. Medium size. Moderate cost.	Less voltage stresses. Common ground for input and output. Less EMI influences.	Lack of extendable input. Generates common and differential mode noises. VMCs blocked the voltage stress across switches.	[35]
High gain Z-Quasi Resonant DC-DC Converter	Individual energy glide manipulates.	High gain with low duty cycle. No isolation. Moderate cost.	High gain ZQR DC/DC Converter with multiple EV. Optimized control strategy for charging and discharging of battery. Winding conduction loss is more evident in maximum gain.	[36]
Non-Isolated High- Gain Triple Port DC- DC Converter	Medium complexity. Positive voltage polarity is maintained. Medium size.	Integrating with bidirectional converter. Simple working. Higher conversion gain. Simple circuitry arrangements.	It provides robust option for interfacing multiple renewable energy sources. Two unidirectional and one bidirectional power flow.	[45]
Non-Isolated DC-DC converter with High voltage gain and Zero ripple current	Medium complexity. Low voltage stress on switches. Input current ripples is almost zero with constant DC value.	High voltage gain. Achieving one voltage gain for whole range of duty cycle. Voltage regulation controller is simpler. Low cost on implementation of driver modules.	Number of inputs can be extended. Provided power of each converter with the extracted efficiency for the operating power rating.	[58]
Multilevel High Step- Up DC-DC Converter	Less components compared to counterparts. Medium size.	High output voltage-gain without using high duty rates. Absorbing energy simultaneously from several input sources. Simple PWM switch control method.	N number of inputs and M-level of converters can be extended. Diode-capacitor cell is used to enhance output voltage level. Current stress on each input voltage is reduced.	[59]
A Novel Step Up Multi-input DC DC Converter for Hybrid Electric Vehicles Application	High energy density. Temperature performance is high. Recyclable of Li-ion batteries.	In the absence of one or two resources, the converter can supply the required power to the load.	Since initial cost of PVs is high.	[80]

TABLE 13. (Continued.) Comparison of non-isolated DC-DC converter topologies.

High Gain SIDO DC- DC Converter	Voltage gain is 4 times higher than conventional boost converter. Continuous input current. Switches and diodes stress Voltages are low.	High gain voltage employs same gate pulse for two switches. Easy to control. Elements count is less.	Number of outputs can be extended. Switched capacitor network is used for high voltage gain.	[85]
Cost-efficient non- isolated three-port DC-DC converter	Handled diversified energy sources of different voltage. Delivering twice the input power from battery. TPB ² C is better performance.	The converter possesses a condensed configuration with fewer components and can provide power to the load.	The primary drawback is that a problem with one converter component will have an impact on the output of the entire converter system.	[116]
High step-up Transformerless DC- DC converter	Less component counts. Medium size.	Greater adaptability compared to analogue electronics. Ease of use. Reliability and stability will improve.	Expandable MIMO converter with high performance. Integration of energy source and deliver to multiple loads. Normalized low power stress on switches.	[117]
MIMO DC-DC converter without Energy Buffer Stages.	Combining of pulsating source cell and output filter cells.	Fewer passive components. Design, structure, and implementation is easy. Without energy buffer stages.	Low-power IoT applications. Generate various output voltages and distribute input power from multiple energy sources.	[121]
New High Step-Up MIMO DC-DC Converter	High output voltage levels. Diode capacitor VM stages can be utilized.	The capacity to utilize various input power sources. Being able to work at low and high-power levels. continuous current input	Simultaneously use the input source with different range of power feeding their output loads.	[126]
A Digital Method of Power-Sharing and Cross-Regulation Suppression for SI- MIMO DC–DC Converter	Variable switching frequency. Online optimization results in fast and dynamic response. Ability of power-sharing & cross-regulation suppression.	Load regulation, line regulation, response to output voltage reference step change, and response to input current reference step change make up the dynamic performance verification.	In continuous conduction mode, the cross- regulation issue with SIMO single inductor dc- dc converters exists.	[130]

an error signal derived from comparing the desired output voltage (reference voltage) with the actual output voltage.

The duty cycle determines the ON-time and OFF-time of the switching element, which, in turn, controls the average output voltage. By continuously adjusting the duty cycle based on the error signal, PWM control ensures stable and accurate output voltage despite changes in input voltage or load conditions. PWM control is widely used in various applications, including power supplies, motor drives, LED lighting, and more, due to its efficiency and versatility. In conclusion, implementing PWM for a DC-DC converter involves using a high-frequency switching element to regulate the output voltage by adjusting the duty cycle of the switch. This technique provides efficient voltage conversion, precise voltage regulation, and flexibility in adjusting output levels. To effectively translate and regulate DC voltage levels for various applications, it is frequently employed in electronics. Fig. 9 shows the basic structure of a multi-input multi-output converter.

B. HYSTERESIS CONTROL

A comparator generally changes its output states when the voltages between its inputs cross through approximately zero



FIGURE 9. PWM control techniques.

volts. However, the noise can cause voltage variations at the inputs, resulting in very quick changes between the two output states. Such unexpectedly quick changes may cause the converter to switch on or off unintentionally and at the wrong times. A small amount of hysteresis is added to the comparator to prevent this undesired output oscillation. A hysteresis comparator has two switching points: one switching point for rising voltages and another for falling voltages are present in a hysteresis comparator. A comparator gains hysteresis by including positive feedback. Hysteretic control (HC) operates



FIGURE 10. Hysteresis control techniques.

based on the concept of maintaining the output voltage within a specific hysteresis band centred around the desired set-point voltage of the converter. This method gives a quick reaction by avoiding the usage of slow feedback found in alternate methods. It also enables more configuration flexibility without any forced constraints. The duty cycle can be changed in the full range from 0 to 1. The dynamic control properties of hysteretic control allow for a reduction in the size of output filters, leading to decreased overall costs of DC-DC converters. Fig. 10 illustrates a schematic representation of a DC-DC converter employing the Hysteretic Control (HC) method [176]. The extent of the hysteresis range and any overshoot, which can be brought on by signal delays and the converter's structural properties, affect the amount of ripple in the output voltage.

In conclusion, implementing hysteresis control for a DC-DC converter is a simple and responsive feedback technique used in DC-DC converters. By comparing the output voltage to upper and lower thresholds (hysteresis band), it controls the output voltage and rapidly modifies the switching element in response to these thresholds. With this method, rapid voltage regulation, less ripple, and resistance to change are all guaranteed. However, it might not be precise enough and needs careful planning to prevent instability. Low-power converters for mobile devices and scenarios requiring prompt and accurate voltage regulation frequently employ hysteretic control.

C. PROPORTIONAL INTEGRAL DERIVATIVE CONTROL

The proportional integral derivative (PID) control is the most commonly used control technique in various industries and has gained universal acceptance for numerous applications, including renewable energy and motor drive systems. The preference for PID control over other techniques stems from its robust response across a wide range of operating conditions and its simple functionality, as demonstrated in Table 14. Despite being a conventional and effective control method, PID is still widely integrated into numerous applications, and one such example can be observed in Fig. 11, where PID control is employed to regulate the DC-DC converter. The PID control method works by receiving data from the DC-DC converter's output and modifying the input switching duty cycle, which controls gain. In many cases, this method delivers minimal efficiency. When compared to other control systems, PID has the additional benefit of having a low



FIGURE 11. PID control with limitation techniques.

control complexity. In addition, there are ongoing initiatives to improve control and effectiveness in renewable energy applications [161], [162] by fusing PID with contemporary control methods, creating hybrid control systems that take advantage of the advantages of both methodologies. This hybridization is proving to be beneficial in achieving improved performance and efficiency in renewable energy systems.

In conclusion, one common method for controlling the output voltage or current of DC-DC converters is to use PID controllers. Proportional (P), integral (I), and derivative (D) terms are its three constituent parts. Current error is addressed by the P term, steady-state errors are eliminated by the I term, and future error changes are anticipated by the D term. PID controllers that have been properly tuned provide precise, reliable, and rapid regulation, enabling them to adapt to various conditions. Tuning complexity, noise sensitivity, and possible nonlinearities are difficulties. Despite these difficulties, PID control is frequently employed in DC-DC converters due to its adaptability and efficiency in establishing precise and responsive control.

D. MODEL PREDICTIVE CONTROL

The model predictive control (MPC) method falls under the category of non-conventional control techniques. Model predictive control techniques can be divided into two main categories: Continuous Control set MPC (CCS-MPC), and Finite Control-States set MPC (FCS-MPC). In CCS-MPC, a modulator generates switching signals from the continuous output of the predictive controller, yielding a constant switching frequency. On the other hand, FCS-MPC provides a benefit by addressing the optimization problem by using a constrained set of switching states.

The main advantages of FCS-MPC are a decreased computing load and the direct application of control actions to the converter without the requirement for a modulation stage [154]. The control algorithm is derived from the system's model to forecast future values of the controlled system. In renewable energy applications [155], MPC is infrequently mentioned as a means for coordinating multiple non-isolated DC-DC converters. MPC operates through a feedback loop that employs a control algorithm and based on increased stability for DC bus voltage for supplying loads is considered in [156]. However, virtual synchronous control is still in the promotion stage and requires to be further explored for the undesired transient instability and overcurrent. Power electronic converter-based MPC, including continuous control set MPC (CCS-MPC) and finite control set MPC

Control Technique	Advantages	Limitations	Features	Ref
Pulse Width Modulation (PWM)	PWM provides more efficient controlling power compared to linear control methods. precise control over the output voltage. It can help in reducing electromagnetic interference (EMI)	Losses can reduce overall efficiency, especially at higher switching frequencies. Design and tuning a feedback control system can be complex and require careful consideration of stability and dynamic response.	It can be easily adapted to different input and output voltage levels. The rapid switching action helps in reducing high-frequency harmonics that contribute to EMI	[178]
Model Predictive Control (MPC)	Effective tracking using strategies based on estimation. Performance optimization for transients under external limitations. Quick response.	High computational. Detailed model should be known. Circuit parameters are sensitive	Iterate online is easy. Non-linear control and robust. Future state predictions.	[157]
Proportional Integral Derivate (PID)	Implementation is simple. Transient response is faster. Integration with numerous control techniques is easy.	It cannot respond to External disturbance load variation Overshoot, steady state error, and settling time is high.	Easy implement. Complexity is low. Suitable for linear control.	[161]
State Space Modeling (SSM)	Transient response will be improved. Load varying with less overshoots.	Detailed model should be known. Initial implementation requires more time.	It is suitable for MIMO systems. Non-linear control.	[163, 164]
Sliding Mode Control (SMC)	Implementation is simple. It can respond for external disturbance.	Chattering problems due to discontinuous control law. High overshoots.	It has robust control. Converge towards sliding surface.	[165, 166, 179]
Fuzzy Logic Control (FLC)	Less overshoots. Tracking response is efficient. Mathematical model is not required.	High computational burden. Settling time is high. It requires rule's for operation.	Robust and non-linear control. Stability over large variations. Suitable for systems having vague boundary conditions.	[168, 170, 174]

TABLE 14. Characteristics of control techniques.

(FCS-MPC), is used to determine optimal switching states of semiconductor switches at the device level to achieve AC/DC, DC/DC or AC/AC power conversion and to stabilize voltages, currents, and different Master-Slave controllers were proposed in [157]. Similarly, more advanced predictive control with new mathematical formulations and holistic intelligent schemes plays a significant role in microgrids, especially in areas like DC microgrids and networked microgrids. The predictive control approach starts with the basic principle and it is applied in three control layers of microgrid hierarchical control. Additionally, the benefits of flexible integration of objectives, faster dynamic responses, challenges and limitations are analysed in [158]. Direct MPC is used to optimize the power flow between the energy sources and the motor. However, the designed MPC may produce better results because it is inherently more robust & better at disturbance rejection [159]. Furthermore, [160] improved the MPC by introducing a multi-level optimization approach in the aerodynamic prediction model of a flapping-wing.

As seen in Table 14, MPC demonstrates the ability to control multi-input, multi-output (MIMO) systems. The fact that it is a multivariable controller, which successfully manages numerous outputs and interactions between system variables at once, is its greatest advantage. MPC also contributes significantly to several other areas, including considerable cost function minimization, operational cost reduction, economic load dispatch, and optimized power flow management. Controlling a DC-DC converter is a critical task, even though the converter topology is highly efficient. The control technique



FIGURE 12. MPC control techniques.

must be responsive to the intermittent changes and uncertain behaviour that may occur in the DC-DC converter. The contemporary predictive control technique (MPC) is implemented for the DC-DC converter, as shown in Fig. 12. The control method uses the algorithm to forecast future values using the output feedback from the DC-DC converter. The control process is improved by this prediction capability, which results in a more seamless output operation. It is significant to note that changes related to the load being handled may occur at the input side as well as the output side of the DC-DC converter.

MPC can perform better than conventional control systems for DC-DC converters, especially when there are restrictions or disruptions. Additionally, it can be utilized to reduce energy consumption or improve the converter's performance under various operating conditions.



FIGURE 13. SSM control techniques.

In conclusion, to implement MPC for a DC-DC converter, we must:

a. Create a system model using state space modelling.

b. Create a cost function that measures how well the control technique performs.

c. To determine the ideal control action, solve an optimization problem at each time step.

d. Have the DC-DC converter carry out the control function.

E. STATE SPACE MODELLING CONTROL

State Space Modelling (SSM) is a mathematical approach used to represent a physical system using a set of inputs, outputs, state variables, and differential equations. It involves two types of equations known as the state equations. The order or complexity of an SSM, i.e., the number of differential equations required to represent a physical model, is determined by the number of input and output variables present in the corresponding physical system [163], [164]. By decomposing complex systems into state variables and the corresponding equations, SSM offers a systematic and thorough technique to characterize the dynamics and behaviour of complex systems. Numerous engineering disciplines use this modelling technique extensively because it enables the investigation, control, and optimization of dynamic systems. Using a time domain technique, it can easily depict higher-order physical systems. SSM-based control may analyse any non-linear system with many inputs and outputs since fundamental representations of physical systems are a prerequisite [191].

In Fig. 13, the implementation of the state space modelling technique for controlling the operation of the DC-DC converter is illustrated. State space modelling is a mathematical control technique that employs a mathematical model to regulate different states of the system. One of its key advantages is its higher efficiency compared to other control techniques. Additionally, it proves to be very effective when iterating parameters, facilitating fine-tuning for optimal performance. State space modelling stands out for its potential to reduce system complexity, which cuts down on calculation time. It streamlines control and enables quicker and more effective computations by simplification the system's representation.

This trait is especially useful in situations where a prompt or real-time response is essential.

The state space model can be used to analyse the behaviour of the DC-DC converter under different operating conditions, such as changes in the input voltage or load current. It can also be used to design control strategies to regulate the output voltage of the converter.

In conclusion, to model the state space of a DC-DC converter, we must:

a. List the system's state variables, including the output voltage, inductor current, and capacitor voltage.

b. Create the equations that explain the state variable dynamics.

c. Combine the equations to create a group of first-order differential equations, which represent the system's state space.

d. Analyse the converter's behaviour and create control plans, using the state space model.

F. SLIDING MODE CONTROL

A non-linear discontinuous control method is sliding mode control (SMC). Even in the presence of external shocks and uncertainty, SMC can function properly. To reduce the error, the system converges to the sliding surface and seeks to glide on the sliding surface for continuous stability. During this process, an error known as the chattering effect is noticed [165], [166], [167]. The chattering effect is the oscillation that can be seen before the system tends to operate in constant sliding mode. SMC is combined with a variety of additional control methods to combat the chattering fault. Drive control, RES generation, and electric vehicles are just a few of the industries that can benefit from adaptive and reliable slide mode control technology. Maintaining output values that are equivalent to or nearly like the values assigned on the sliding surface is the goal of slide model-based control. In Fig. 14, the Sliding Mode Control (SMC) technique is integrated with the DC-DC converter topology to achieve optimum operation even during varying load conditions. The feedback from the system is fed into the SMC, which uses this information to perform iterations and determine the appropriate switching operation of the DC-DC converter. Sliding Mode Control is a robust control technique that aims to keep the system on a predefined sliding surface, ensuring stability and accurate tracking of desired outputs. By continuously adjusting the control inputs based on the system's state and reference trajectory, SMC can handle uncertainties and disturbances, making it suitable for applications with changing load conditions. By integrating SMC with the DC-DC converter, the system can adapt and respond effectively to load variations, ensuring stable and efficient operation under different circumstances. An integral terminal term is considered in the proposed control approach to increase convergence time and decrease steady-state tracking error [184]. The suggested controller has excellent transient stability performance during operating point changes such as dramatic shifts in load demands, solar irradiance fluctuations, and so on.



FIGURE 14. SMC control techniques.

Sliding Mode Control (SMC) is a robust technique used in DC-DC converters for regulating output voltage or current. It guides the system onto a predefined sliding surface in the state space, where dynamics maintain stability despite uncertainties and disturbances. SMC consists of switching and reaching control terms, ensuring accurate regulation and fast responses. It excels in uncertain conditions and doesn't rely heavily on precise modelling. However, challenges include the potential for chattering (high-frequency oscillations) and the need for skilful design. SMC offers resilience, adaptability, and rapid control in challenging scenarios, making it a valuable choice for DC-DC converter regulation.

In conclusion, to model the sliding mode control of a DC-DC converter, we must consider:

a. A reliable control method for managing the output of power converters (such as DC-DC and AC-DC) is sliding mode control. Creating a sliding surface that depicts the required system behaviour is necessary.

b. The control law that consists of reaching and sliding control terms to drive the system state to the sliding surface and maintain it there.

c. The system enters a sliding mode while it is on the sliding surface, making it immune to uncertainties and disturbances.

d. The main challenge is dealing with chattering, high-frequency oscillations that may arise during sliding mode.

e. Sliding mode control is appropriate for applications with uncertain or changing parameters because it gives resistance to parameter fluctuations and external disturbances.

G. FUZZY LOGIC CONTROL

Fuzzy Logic Control (FLC) belongs to the category of nonconventional and non-linear control techniques. It operates based on a heuristic approach that imitates the human thought process. To implement this thought process, FLC requires predefined rules. Linguistic rules are established for the input and output of the system, represented as membership functions. Compared to State Space Modelling (SSM), FLC is thought to be somewhat simpler because it doesn't call for an accurate mathematical description of the system. It also has nonlinearity handling capabilities, like Model Predictive Control (MPC) and SSM. The unique control process of FLC involves taking feedback in the form of crisp values from the system, converting it into linguistic form, and then comparing it with the membership functions, a process commonly known as fuzzification. FLC is suitable for systems with ambiguous or imprecise input because it can produce control decisions that are easier to understand and intuitive by employing these linguistic norms. After fuzzification, Fuzzy Logic Control (FLC) converts the linguistic form back into a crisp value, which is known as defuzzification. This process allows FLC to produce a precise and actionable control output based on fuzzy rules and linguistic variables. FLC does provide effective solutions, especially when it comes to tracking behaviour in non-linear control systems that may have confusing boundary conditions. It is ideal for a variety of real-world applications due to its capacity to manage ambiguous and imperfect information. However, like any control technique, FLC also has its drawbacks. One of the main concerns is the higher computational time, which can limit its real-time applicability in some cases. To address this issue, FLC is often integrated with other control techniques and used in offline mode [168]. By combining the strengths of FLC with other control methods, engineers can leverage the advantages of both approaches while mitigating the computational time limitations of FLC as illustrated in Table 14. This hybrid approach allows for improved performance and efficiency in control systems that require both accuracy and real-time responsiveness.

FLC is used in both household and industrial applications as a reliable control method. The autonomous automobile braking system and the charge control in electric vehicles also use FLC in the automotive industry [169], [170]. Additionally, it is used to monitor and direct maritime surface vessels and undersea vehicles [171]. Fuzzy Logic Control (FLC) is a versatile and effective control technique that finds applications in various domains, including both domestic and industrial settings. In the automotive sector, FLC is employed in advanced driver-assistance systems (ADAS), particularly for automatic car braking systems. By using FLC, the braking system can make real-time decisions based on inputs from sensors and the driving environment, leading to enhanced safety and collision avoidance. FLC is also utilized for charge control in electric vehicles, helping to optimize battery charging and discharging processes, thereby extending battery life FLC has also opted for industrial applications and the power generation sectors improving overall efficiency [173]. [172], [173], [174]. Multi-input non-isolated DC-DC converters used in power generation have FLC applied to them so they can run in a boost mode under a grid-connected scheme [175]. With the help of this application, renewable energy sources can be integrated into the grid and converted to power efficiently. The control design of a DC-DC converter using a fuzzy logic controller is shown in Fig. 15. FLC is a useful tool in a variety of applications because of its adaptability and capacity to manage challenging and uncertain circumstances, offering effective and intelligent control solutions for numerous systems and processes.

H. MAXIMUM POWER POINT TRACKING CONTROL

A solar panel or any other renewable energy source with variable output characteristics, like wind turbines, can be



FIGURE 15. FLC Control Techniques.



FIGURE 16. (a) Block diagram of the controlled MPPT system for DC-DC converter, (b) Scheme of the control system including the MPPT.

controlled to provide the most power possible using the MPPT approach. The goal of MPPT is to position the DC-DC converter so that it operates at the point on the IV curve, or voltage-current curve, of the source where the most power is being generated. With MPPT, power extraction from PV modules is optimized for a variety of environmental factors, particularly solar irradiance and temperature. By lining up the PV modules' Maximum Power Point (MPP) with the converter's operational voltage and current, this optimization is made possible. The fundamental block diagram of a typical PV system with MPPT and control scheme system including MPPT is shown in Fig. 16 (a -b). The DC-DC converter can be replaced with an inverter, which can then be connected to the utility grid, to transform the stand-alone DC-DC system seen in the diagram into a grid-connected system. The MPPT operates in the manner described below. An initial current and voltage sensor measures the current and voltage of the PV array. These numbers are passed into an MPPT block, which computes the MPP for that specific sampling cycle. The comparative chart of control techniques is illustrated in Table 15.

MPPT is crucial because it enables the system to continue operating at the MPP despite changes in the external environment. MPPT makes sure that the load receives the most power from the source by continuously altering the duty cycle or switching frequency of the DC-DC converter. It is necessary to use a maximum power point tracking (MPPT) controller to make fuel cell systems more effective. This study [186] proposes an effective approach for MPPT of proton exchange membrane (PEM) fuel cells based on particle swarm optimization (PSO) and PID controller (PSO-PID). The Perturb and Observe (P&O), Incremental Conductance, and Fractional Short-Circuit Current approaches, among others, are some of the MPPT algorithms that are used to do this. These algorithms keep track of the output power, voltage, or current and modify the DC-DC converter's operating point as necessary to maintain the MPP. To increase energy harvest and boost overall system efficiency, MPPT techniques are frequently utilized in solar power systems and other renewable energy applications. In such DC grid applications, a DC/DC converter is essential for interconnecting DC buses with different voltage levels, which needs to meet several requirements, including bidirectional power conversion capability, high power rating, high step-ratio, high power density, and high efficiency. A modified MPC MPPT [187] is based on a one-voltage sensor, one-current sensor, and one-output voltage observer.

In conclusion, MPPT maximizes energy conversion efficiency and overall energy yield by keeping operations close to the MPP and optimizing DC-DC converters for renewable energy systems.

1) PERTURBATION AND OBSERVATION (P&O)

The perturb and observe approach is straightforward, requires no prior understanding of the features of PV generators, or the measurement of solar intensity and cell temperature, and is simple to implement with analogue and digital circuitry. Even though the solar irradiance and cell temperature are constant, it disturbs the system's operating point, causing the terminal voltage of the PV array [182] to oscillate around the MPP value. Due to its ability to strike a compromise between performance and simplicity, it is also the most used and reliable MPPT algorithm. However, it struggles with a lack of speed and adaptability, both of which are essential for monitoring quick transients in a variety of environmental circumstances. In [181], fixed zone perturbs and observations (FZPO) are used to achieve efficiency, fast, and drift-free MPPT for PV systems. In the FC system, the MPPT controller, such as the perturbation and observation (P&O) is operated by the P-V curve of FCs. Finally, the activation voltage drop, ΔV_{act} can be expressed as follows [180]: where ϵ_i with i = 1, 2, 3,4 represents the cell's parametric coefficient and Co2 is the dissolved oxygen concentration.

2) INCREMENTAL CONDUCTANCE (INCCOND)

The P&O approach has the two aforementioned significant drawbacks. (1) A minor power variation around the MPP is always present due to a constant amount of disturbance at the steady state, which adds to some power losses. (2) The

Control Technique Design and Implementation		Operating Point	Reaction	Settling Time
Hysteresis Control (HC)	Design and implementation are simple. It doesn't required feedback.	The operating point variations and load disturbances are responds directly.	HC is free from disturbances. It provides stability against transient condition. Fast respond in dynamic responses.	Settling time is higher that MPC controller
Model Predictive Control (MPC)	Predictive 1 (MPC)Implementation has been restricted to slowly varying systemssuitability for directly addressing multi-variable systems.MPC desaturation capabilities, without relying.Controller essentially linear.It is still with tangible benefits.		Settling time is lower than FLC.	
Proportional Integral Derivate (PID)	Design is simple. Self-governing model.	It won't react on operating point load disturbances. It operates directly on error signals.	PID cannot meet increasing necessities for fast dynamic response. High control accuracy.	Settling time is higher than SMC.
State Space Modeling (SSM)	Simple in design	Respond directly on large load disturbance	Better dynamic response.	Settling time is lesser than MPC.
Sliding Mode Control (SMC)	Execution and propose is extremely simple.	Unsettling influences are burden. Working point varieties react appropriately.	SMC is liberated from unsettling influences, and requirements. Lastingness against enormous aggravations. Unique reaction is quick.	Minimal settling time.
Fuzzy Logic Control (FLC)	Design and implementation are quite simple.	The operating point variations and load instability are responded directly	FLC is free from disturbances. It provides stability against large disturbances. Fast respond in dynamic response.	Settling time is higher than PID controller.

TABLE 15. Comparative chart of Control Techniques [177].

operating point is most likely to deviate from the genuine MPP in environments with fast environmental changes. To get around the drawbacks, the INC technique was suggested. Therefore, incremental resistance (ICR) rather than INC will be used to define the algorithm. The INC algorithm [189] is dependent on the perturbation value's "offset," just like the P&O algorithm. The large value of 'offset' will result in fasttracking, but the system may oscillate at MPP. Instantaneous inductance (I/V) and incremental conductance (I/V) are not identical. In a double-stage configuration, the front-end DC-DC converter is used to extract the maximum power point tracking (MPPT) [174] power from the SPV array by using additional control. However, this issue can be solved with a smaller value of "offset," but tracking will become slower. As in the case of the P&O approach, the trade-off still exists. The fixed and adaptive values of "offset" can also be used to characterize two subgroups of the INC technique. Because they share the same purpose as the P&O approach, they are not described individually. Since INC and P&O are based on the same idea. INC is also unable to track the GMPP.

3) SOFT COMPUTING MPPT

For simplicity, they are grouped into several categories, and it is worth noting that for SC-based MPPT, not all techniques are used as MPPT controllers. Artificial Neural Network (ANN) is sometimes utilized to optimize certain parameters of other MPPT controllers [180], e.g. Fuzzy Logic Controller (FLC) [185], P&O [188] and HC. Furthermore, it is common to combine two or more techniques in a single MPPT controller. For example, ANN can be combined with FLC, or PSO with P&O. In most cases, the performance of the hybrid tends to be superior compared to a stand-alone SC method. Table 16 illustrates the comparison of different soft computing techniques. In Table 16, the standards for "fast" and "very fast" convergence are defined and differentiated based on the number of iterations required for convergence. For instance, a method may be categorized as having "fast" convergence if it typically converges within a moderate number of iterations in one minute, while a method classified as "very fast" convergence would converge in ten seconds significantly for fewer iterations.

VI. CHALLENGES AND FUTURE TRENDS

Recent research has extensively explored the modelling, design, and use of Multiport Converters across various applications, including hybrid generation systems, electric vehicles, fuel cells, uninterruptible power supplies, and microgrid telecom power systems.

A few key technical issues identified in Multiport converters are:

- Utilization of single energy sources at a given time, necessitates time-sharing concepts to mitigate power coupling effects, though imposing constraints on energy source utilization and output voltage due to duty ratio limitations.
- Limitations of existing solutions such as voltage dividers, lead to losses in passive components.
- Computational time and processing complexity emerge as significant challenges in addressing these technical issues and research gaps.

Methods	PV array dependency	Convergence	Partial shading	Hardware implementation	Algorithm complexity
Fuzzy Logic Control (FLC)	Yes	Fast	Standalone – No coupled – yes	Easy	Moderate
Differential Evolution (DE)	No	Varies/ Fast	Yes	Easy	Moderate
Particle Swarm Optimization (PSO)	No	Fast	Yes	Moderate	Moderate
Ant Colony Optimization (ACO)	No	Fast	Yes	Moderate	Moderate
Artificial Neural Networks (ANN)	Yes	Fast	Standalone – No coupled – yes	Difficult	Moderate
Genetic Algorithm (GA)	No	Fast	No	Easy	Difficult
Bayesian network	No	Very fast	Yes	Difficult	Difficult
Chaotic search	No	Fast	Yes	Difficult	Very difficult
Non-linear predictor	No	Fast	No	Easy	Easy

TABLE 16. Comparison of soft computing techniques.

These converters are useful for integrating diverse energy sources with varying power or voltage levels to achieve regulated output voltage. These converters follow specific operational conditions: a single energy source feeds the load at a given time, and time-sharing concepts are employed to avoid power coupling effects. However, this approach restricts energy source utilization and output voltage due to duty ratio constraints. By reducing operating time delays and implementing lower duty ratios, converter size can be reduced while maintaining output voltage quality, especially for applications requiring high voltage gain. Ensuring fault tolerance for alternate sources and switches is crucial in Multiport converter design.

A persistent challenge is the optimal control for both input and output voltages, particularly in cases requiring multiple output voltages. Existing solutions like voltage dividers cause losses in passive components. A proposed solution is creating new multi-input topologies that efficiently provide both DC and AC output voltages, reducing installation cost and size. Some efforts are also needed to enhance the optimal control of multiple outputs. While many converters can handle wide input voltage ranges, improvements are sought, especially for drawing energy from very low renewable energy sources. Future research should target circuit simplification, component reduction, enhanced voltage gain, and power scalability. These advances would lower costs, improve output, and contribute to the transition to renewable energy, making it imperative to address reliability, efficiency, and power loss issues in multi-input converters.

The evolution of hybrid control techniques is highlighted, comparing various methods in Table 14 based on features, advantages, and limitations. Factors such as response time, system complexity for higher-order systems, stability, and overshoots are considered. The simplest method, PID control requires basic parameter tuning but struggles with fast dynamic response, input imbalances, and optimization. SMC stands out for stable DC-DC converter operation during disturbances and rapid responses. However, it demands high switching frequencies, leading to efficiency loss. Multiport Converters are tailored to converter characteristics, and reduce losses by controlling duty cycles but are complex, especially with non-linear models. SSM excels in accuracy and robustness by aligning with converter energy equipment, accommodating nonlinearity easily. Fuzzy logic modern technique excels in tracking nonlinear systems with membership functions resembling human thinking, suitable for renewable energy control. Yet, it faces challenges of computational time and processing complexity.

VII. CONCLUSION

This review paper delved into the performance analysis of various non-isolated DC-DC converter topologies, emphasizing their application in renewable energy and hybrid vehicles. The advantages and drawbacks of multi-input converters, particularly in terms of reduced integration costs and improved reliability have been discussed. Various control techniques and ongoing efforts to enhance the efficiency and cost-effectiveness of multi-input converters have also been

discussed. Multi-port converters in various applications are also discussed. From this review, the following has been observed and concluded.

- Non-isolated DC-DC converter topologies, including MISO, SIMO, and MIMO, offer distinct advantages in renewable energy and hybrid vehicle applications, with multi-input converters showing promise due to reduced integration costs and enhanced reliability through lossless soft switching circuits.
- Three-port converters, evolving towards multi-inputmulti-output configurations, are gaining traction in renewable energy systems, particularly for their ability to optimize power flow.
- Control techniques such as PWM, Hysteretic, and MPC play a crucial role in voltage regulation and power transfer efficiency, with digital control advancements offering adaptability and sophistication.
- MPPT algorithms, notably P&O and Incremental Conductance, are pivotal in maximizing power generation efficiency in diverse converter types, while ongoing research explores complex soft computing methods.
- Multi-port converters find extensive application across hybrid generation systems, electric vehicles, microgrids, and telecommunications, facilitating efficient power management and integration of multiple energy sources.
- Challenges including reducing operating time delays, implementing lower duty ratios, and optimizing control for multiple output voltages are areas of focus for researchers, aiming to advance circuit simplification, component reduction, and power scalability in multiinput converters.
- The comparison of hybrid control techniques such as PID, SMC, MPC, and fuzzy logic provides valuable insights into choosing appropriate controllers based on factors like response time, stability, and complexity.

Overall, this review provides a comprehensive understanding of efficient multi-input converter topologies for both DC and AC output voltages, guiding researchers towards lowering installation costs and sizes while enhancing performance and functionality.

REFERENCES

- A. Sinha and M. Shahbaz, "Estimation of environmental Kuznets curve for CO₂ emission: Role of renewable energy generation in India," *Renew. Energy*, vol. 119, pp. 703–711, Apr. 2018, doi: 10.1016/j.renene.2017.12.058.
- [2] H. Sarnago, Ó. Lucía, M. Pérez-Tarragona, and J. M. Burdío, "Dual-output boost resonant full-bridge topology and its modulation strategies for high-performance induction heating applications," *IEEE Trans. Ind. Electron.*, vol. 63, no. 6, pp. 3554–3561, Jun. 2016, doi: 10.1109/TIE.2016.2530780.
- [3] G. Chen, Y. Deng, J. Dong, Y. Hu, L. Jiang, and X. He, "Integrated multiple-output synchronous buck converter for electric vehicle power supply," *IEEE Trans. Veh. Technol.*, vol. 66, no. 7, pp. 5752–5761, Jul. 2017, doi: 10.1109/TVT.2016.2633068.
- [4] S. Naresh, S. Peddapati, and M. L. Alghaythi, "A novel high quadratic gain boost converter for fuel cell electric vehicle applications," *IEEE J. Emerg. Sel. Topics Ind. Electron.*, vol. 4, no. 2, pp. 637–647, Apr. 2023, doi: 10.1109/JESTIE.2023.3248449.

- [6] B. Ullah, H. Ullah, and S. Khalid, "Direct model predictive control of noninverting buck-boost DC–DC converter," *CES Trans. Electr. Mach. Syst.*, vol. 6, no. 3, pp. 332–339, Sep. 2022, doi: 10.30941/CES-TEMS.2022.00043.
- [7] M. D. M. Vieira and M. A. J. Huijbregts, "Comparing mineral and fossil surplus costs of renewable and non-renewable electricity production," *Int. J. Life Cycle Assessment*, vol. 23, no. 4, pp. 840–850, Apr. 2018, doi: 10.1007/s11367-017-1335-6.
- [8] F. Martins, C. Felgueiras, M. Smitkova, and N. Caetano, "Analysis of fossil fuel energy consumption and environmental impacts in European countries," *Energies*, vol. 12, no. 6, pp. 964–975, Mar. 2019, doi: 10.3390/en12060964.
- [9] F. Perera, "Pollution from fossil-fuel combustion is the leading environmental threat to global pediatric health and equity: Solutions exist," *Int. J. Environ. Res. Public Health*, vol. 15, no. 1, p. 16, Dec. 2017, doi: 10.3390/ijerph15010016.
- [10] M. H. Qais, H. M. Hasanien, and S. Alghuwainem, "Optimal transient search algorithm-based PI controllers for enhancing low voltage ride-through ability of grid-linked PMSG-based wind turbine," *Electronics*, vol. 9, no. 11, pp. 1807–1827, Oct. 2020, doi: 10.3390/electronics9111807.
- [11] M. A. Soliman, H. M. Hasanien, and A. Alkuhayli, "Marine predators algorithm for parameters identification of triple-diode photovoltaic models," *IEEE Access*, vol. 8, pp. 155832–155842, 2020, doi: 10.1109/ACCESS.2020.3019244.
- [12] A. Colmenar-Santos, M. Monteagudo-Mencucci, E. Rosales-Asensio, M. de Simón-Martín, and C. Pérez-Molina, "Optimized design method for storage systems in photovoltaic plants with delivery limitation," *Sol. Energy*, vol. 180, pp. 468–488, Mar. 2019, doi: 10.1016/j.solener.2019.01.046.
- [13] L. Hernández-Callejo, S. Gallardo-Saavedra, and V. Alonso-Gómez, "A review of photovoltaic systems: Design, operation and maintenance," *Sol. Energy*, vol. 188, pp. 426–440, Aug. 2019, doi: 10.1016/j.solener.2019.06.017.
- [14] S. T. Meraj, N. Zaihar Yahaya, B. S. M. Singh, and R. Kannan, "Implementation of a robust hydrogen-based grid system to enhance power quality," in *Proc. IEEE Int. Conf. Power Energy (PECon)*, Dec. 2020, pp. 153–158, doi: 10.1109/PECon48942.2020.9314536.
- [15] M. H. Qais, H. M. Hasanien, and S. Alghuwainem, "A novel LMSRE-based adaptive PI control scheme for grid-integrated PMSGbased variable-speed wind turbine," *Int. J. Electr. Power Energy Syst.*, vol. 125, Feb. 2021, Art. no. 106505, doi: 10.1016/j.ijepes.2020. 106505.
- [16] Y. E. García Vera, R. Dufo-López, and J. L. Bernal-Agustín, "Energy management in microgrids with renewable energy sources: A literature review," *Appl. Sci.*, vol. 9, no. 18, p. 3854, Sep. 2019, doi: 10.3390/app9183854.
- [17] M. Z. Hossain, N. A. Rahim, and J. A. Selvaraj, "Recent progress and development on power DC–DC converter topology, control, design and applications: A review," *Renew. Sustain. Energy Rev.*, vol. 81, pp. 205–230, Jan. 2018, doi: 10.1016/j.rser.2017.07.017.
- [18] K. H. Kumar and G. V. S. K. Rao, "A review of various DC–DC converter topologies for photovoltaic applications," in *Proc. 6th Int. Conf. Commun. Electron. Syst. (ICCES)*, 2021, pp. 49–52, doi: 10.1109/ICCES51350.2021.9489070.
- [19] A. Chauhan and R. P. Saini, "A review on integrated renewable energy system based power generation for stand-alone applications: Configurations, storage options, sizing methodologies and control," *Renew. Sustain. Energy Rev.*, vol. 38, pp. 99–120, Oct. 2014, doi: 10.1016/j.rser.2014.05.079.
- [20] K. S. Reddy, M. Kumar, T. K. Mallick, H. Sharon, and S. Lokeswaran, "A review of integration, control, communication and metering (ICCM) of renewable energy based smart grid," *Renew. Sustain. Energy Rev.*, vol. 38, pp. 180–192, Oct. 2014, doi: 10.1016/j.rser.2014. 05.049.
- [21] Z. Rehman, I. Al-Bahadly, and S. Mukhopadhyay, "Multiinput DC–DC converters in renewable energy applications—An overview," *Renew. Sustain. Energy Rev.*, vol. 41, pp. 521–539, Jan. 2015, doi: 10.1016/j.rser.2014.08.033.

- [22] A. K. Singh, A. K. Mishra, K. K. Gupta, and Y. P. Siwakoti, "High voltage gain bidirectional DC–DC converters for supercapacitor assisted electric vehicles: A review," *CPSS Trans. Power Electron. Appl.*, vol. 7, no. 4, pp. 386–398, Dec. 2022, doi: 10.24295/CPSSTPEA.2022.00035.
- [23] M. S. Bhaskar, M. Meraj, A. Iqbal, S. Padmanaban, P. K. Maroti, and R. Alammari, "High gain transformer-less double-duty-triplemode DC/DC converter for DC microgrid," *IEEE Access*, vol. 7, pp. 36353–36370, 2019.
- [24] P. K. Maroti, S. Padmanaban, M. S. Bhaskar, M. Meraj, A. Iqbal, and R. Al-Ammari, "High gain three-state switching hybrid boost converter for DC microgrid applications," *IET Power Electron.*, vol. 12, pp. 3656–3667, Nov. 2019, doi: 10.1049/iet-pel.2018.6403.
- [25] A. Ghosh and F. Zare, "Review of SISO control systems," in Proc. Recent Adv. Renew. Energy Sources (RARES), 2021, pp. 59–121, doi: 10.1002/9781119815464.ch3.
- [26] K. M. R. Eswar, R. Aravind, C. Bharatiraja, S. Akshat, C. Srishti, and S. Bindu, "Closed-loop control of modified switched inductor boost converter with high voltage gain and reduced switch voltage stress," in *Emerging Technologies in Electrical Engineering for Reliable Green Intelligence* (Lecture Notes in Electrical Engineering). Singapore: Springer, 2024, pp. 1–13, doi: 10.1007/978-981-99-9235-5_1.
- [27] G. Ramanathan and C. Bharatiraja, "Ultra-gain two-input two-output DC-DC converter for electric vehicle application," in *Emerging Technologies in Electrical Engineering for Reliable Green Intelligence* (Lecture Notes in Electrical Engineering). Singapore: Springer, 2024 pp. 159–175, doi: 10.1007/978-981-99-9235-5_12.
- [28] M. Forouzesh, Y. Shen, K. Yari, Y. P. Siwakoti, and F. Blaabjerg, "High-efficiency high step-up DC–DC converter with dual coupled inductors for grid-connected photovoltaic systems," *IEEE Trans. Power Electron.*, vol. 33, no. 7, pp. 5967–5982, Jul. 2018, doi: 10.1109/TPEL.2017.2746750.
- [29] J. Divya Navamani, K. Vijayakumar, and R. Jegatheesan, "Non-isolated high gain DC–DC converter by quadratic boost converter and voltage multiplier cell," *Ain Shams Eng. J.*, vol. 9, no. 4, pp. 1397–1406, Dec. 2018, doi: 10.1016/j.asej.2016.09.007.
- [30] M. Zhou, J. Fu, X. Wu, M. Yang, and Z. Zhang, "A non-isolated high-gain DC/DC converter suitable for fuel cell vehicles," *J. Electr. Eng. Technol.*, vol. 17, no. 1, pp. 271–282, Jan. 2022, doi: 10.1007/s42835-021-00785-7.
- [31] F. Mumtaz, N. Z. Yahaya, S. T. Meraj, N. S. S. Singh, and G. E. M. Abro, "A novel non-isolated high-gain non-inverting interleaved DC–DC converter," *Micromachines*, vol. 14, no. 3, p. 585, Feb. 2023, doi: 10.3390/mi14030585.
- [32] P. K. Maroti, R. Al-Ammari, M. S. Bhaskar, M. Meraj, A. Iqbal, S. Padmanaban, and S. Rahman, "New tri-switching state non-isolated high gain DC–DC boost converter for microgrid application," *IET Power Electron.*, vol. 12, no. 11, pp. 2741–2750, Sep. 2019, doi: 10.1049/ietpel.2019.0236.
- [33] M. S. H. Lipu, S. Ansari, M. S. Miah, K. Hasan, S. T. Meraj, M. Faisal, T. Jamal, S. H. M. Ali, A. Hussain, K. M. Muttaqi, and M. A. Hannan, "A review of controllers and optimizations based scheduling operation for battery energy storage system towards decarbonization in microgrid: Challenges and future directions," *J. Cleaner Prod.*, vol. 360, Aug. 2022, Art. no. 132188, doi: 10.1016/j.jclepro.2022.132188.
- [34] P. Gunawardena, N. Hou, D. Nayanasiri, and Y. Li, "A dual-input singleoutput DC–DC converter topology for renewable energy applications," *IEEE Trans. Ind. Appl.*, vol. 59, no. 2, pp. 1995–2006, Mar. 2023, doi: 10.1109/TIA.2022.3218619.
- [35] S. M. Hashemzadeh, V. Marzang, S. Pourjafar, and S. H. Hosseini, "An ultra high step-up dual-input single-output DC–DC converter based on coupled inductor," *IEEE Trans. Ind. Electron.*, vol. 69, no. 11, pp. 11023–11034, Nov. 2022.
- [36] S. Harini, N. Chellammal, B. Chokkalingam, and L. Mihet-Popa, "A novel high gain dual input single output Z-quasi resonant (ZQR) DC/DC converter for off-board EV charging," *IEEE Access*, vol. 10, pp. 83350–83367, 2022, doi: 10.1109/ACCESS.2022.3195936.
- [37] S. Naeiji, S. E. Abdollahi, S. R. Abdollahi, and B. Baigzadehnoe, "An improved bidirectional non-isolated dual-input DC–DC converter for electric vehicles application," in *Proc. 14th Power Electron., Drive Syst., Technol. Conf. (PEDSTC)*, Babol, Iran, Jan. 2023, pp. 1–7, doi: 10.1109/PEDSTC57673.2023.10087176.
- [38] A. Farakhor, M. Abapour, and M. Sabahi, "Design analysis and implementation of a multiport DC–DC converter for renewable energy applications," *IET Power Electronics*, vol. 12, pp. 465–475, Mar. 2019, doi: 10.1049/iet-pel.2018.5633.

- [39] S. Rostami, V. Abbasi, and N. Talebi, "Ultrahigh step-up multiport DC–DC converter with common grounded input ports and continuous input current," *IEEE Trans. Ind. Electron.*, vol. 69, no. 12, pp. 12859–12873, Dec. 2022, doi: 10.1109/TIE.2021.3131857.
- [40] K. Varesi, S. H. Hosseini, M. Sabahi, E. Babaei, S. Saeidabadi, and N. Vosoughi, "Design and analysis of a developed multiport high step-up DC–DC converter with reduced device count and normalized peak inverse voltage on the switches/diodes," *IEEE Trans. Power Electron.*, vol. 34, no. 6, pp. 5464–5475, Jun. 2019, doi: 10.1109/TPEL.2018.2866492.
- [41] P. Shaw, M. M. Alam, S. Ul Hasan, Y. P. Siwakoti, and D. D.-C. Lu, "A new dual-input single-output step-up DC–DC converter for gridconnected photovoltaic applications," in *Proc. 4th Int. Conf. Smart Power Internet Energy Syst. (SPIES)*, Beijing, China, Dec. 2022, pp. 846–851, doi: 10.1109/SPIES55999.2022.10082645.
- [42] T. Mishra and R. K. Singh, "Analysis and hardware implementation of a single inductor dual input single output DC–DC converter for microgrid applications," in *Proc. Int. Conf. Power, Energy, Control Transmiss. Syst. (ICPECTS)*, Chennai, India, Dec. 2022, pp. 1–4, doi: 10.1109/ICPECTS56089.2022.10047242.
- [43] G. Prakash, K. Harini, and T. Hemapriya, "Analysis of bidirectional dual input single output DC–DC converter for EV application," in *Proc. 9th Int. Conf. Adv. Comput. Commun. Syst. (ICACCS)*, vol. 1, Coimbatore, India, Mar. 2023, pp. 1341–1346, doi: 10.1109/ICACCS57279.2023.10113097.
- [44] M. Veerachary and N. Yadav, "Design and analysis of two-input singleoutput DC–DC converter," in *Proc. IEEE 4th Int. Conf. Comput., Power Commun. Technol. (GUCON)*, Kuala Lumpur, Malaysia, Sep. 2021, pp. 1–6, doi: 10.1109/GUCON50781.2021.9573720.
- [45] B. Chandrasekar, C. Nallaperumal, S. Padmanaban, M. S. Bhaskar, J. B. Holm-Nielsen, Z. Leonowicz, and S. O. Masebinu, "Nonisolated high-gain triple port DC–DC buck-boost converter with positive output voltage for photovoltaic applications," *IEEE Access*, vol. 8, pp. 113649–113666, 2020, doi: 10.1109/ACCESS.2020.3003192.
- [46] R. Faraji, L. Ding, T. Rahimi, M. Kheshti, and M. R. Islam, "Soft-switched three-port DC–DC converter with simple auxiliary circuit," *IEEE Access*, vol. 9, pp. 66738–66750, 2021, doi: 10.1109/ACCESS.2021.3076183.
- [47] J. Wang, K. Sun, C. Xue, T. Liu, and Y. Li, "Multi-port DC–AC converter with differential power processing DC–DC converter and flexible power control for battery ESS integrated PV systems," *IEEE Trans. Ind. Electron.*, vol. 69, no. 5, pp. 4879–4889, May 2022, doi: 10.1109/TIE.2021.3080198.
- [48] M. Dezhbord, P. Mohseni, S. H. Hosseini, D. Mirabbasi, and M. R. Islam, "A high step-up three-port DC–DC converter with reduced voltage stress for hybrid energy systems," *IEEE J. Emerg. Sel. Topics Ind. Electron.*, vol. 3, no. 4, pp. 998–1009, Oct. 2022, doi: 10.1109/JESTIE.2022.3146056.
- [49] S. Kumaravel, V. Karthikeyan, G. G. Kumar, T. J. Jithin, and V. S. Rao, "Dual-input dual-output DC–DC converter for DC microgrid applications," in *Proc. IEEE Int. Conf. Power Electron., Drives Energy Syst. (PEDES)*, Chennai, India, Dec. 2018, pp. 1–6, doi: 10.1109/PEDES.2018.8707726.
- [50] M. Ghavaminejad, E. Afjei, and M. Meghdadi, "Doubleinput/double-output buck-zeta converter," in *Proc. 29th Iranian Conf. Electr. Eng. (ICEE)*, Tehran, Iran, May 2021, pp. 368–372, doi: 10.1109/ICEE52715.2021.9544131.
- [51] R. Aravind, S. Athikkal, R. E. K. Meesala, and C. Bharatiraja, "Analysis of a single inductor based two input two output DC–DC converter," in *Proc. 2nd Int. Conf. Power, Control Comput. Technol. (ICPC2T)*, Raipur, India, Mar. 2022, pp. 1–5, doi: 10.1109/ICPC2T53885.2022.9776681.
- [52] S. Bharadwaj, K. Santha, and S. Arulmozhi, "Design and analysis of integrated dual-input dual-output DC–DC converter for electric vehicle and drives applications," in *Proc. 9th Int. Conf. Electr. Energy Syst. (ICEES)*, Chennai, India, Mar. 2023, pp. 681–686, doi: 10.1109/ICEES57979.2023.10110154.
- [53] K. Suresh, C. Bharatiraja, N. Chellammal, M. Tariq, R. K. Chakrabortty, M. J. Ryan, and B. Alamri, "A multifunctional non-isolated dual inputdual output converter for electric vehicle applications," *IEEE Access*, vol. 9, pp. 64445–64460, 2021, doi: 10.1109/ACCESS.2021.3074581.
- [54] I. Chen, C.-W. Liang, and T.-H. Tsai, "A single-inductor dual-input dual-output DC–DC converter for photovoltaic and piezoelectric energy harvesting systems," *IEEE Trans. Circuits Syst. II, Exp. Briefs*, vol. 66, no. 10, pp. 1763–1767, Oct. 2019.

- [55] H. Shao, X. Li, C.-Y. Tsui, and W.-H. Ki, "A novel singleinductor dual-input dual-output DC–DC converter with PWM control for solar energy harvesting system," *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.*, vol. 22, no. 8, pp. 1693–1704, Aug. 2014, doi: 10.1109/TVLSI.2013.2278785.
- [56] V. Krishnakumar, P. Anbarasan, J. Pradeep, and M. Vijayaragavan, "Modified dual input dual output DC–DC converter for bladeless wind energy harvesting system," in *Proc. 12th Int. Symp. Adv. Topics Electr. Eng. (ATEE)*, Bucharest, Romania, Mar. 2021, pp. 1–6, doi: 10.1109/ATEE52255.2021.9425186.
- [57] Q. Tian, G. Zhou, M. Leng, X. Fan, and T. Yan, "A novel dual-input dual-output converter and dynamic energy management for PV/battery systems," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Baltimore, MD, USA, Sep. 2019, pp. 6468–6472, doi: 10.1109/ECCE.2019.8912649.
- [58] Z. Saadatizadeh, P. C. Heris, X. Liang, and E. Babaei, "Expandable non-isolated multi-input single-output DC–DC converter with high voltage gain and zero-ripple input currents," *IEEE Access*, vol. 9, pp. 169193–169219, 2021, doi: 10.1109/ACCESS.2021.3137126.
- [59] A. Shoaei, K. Abbaszadeh, and H. Allahyari, "A single-inductor multiinput multilevel high step-up DC–DC converter based on switcheddiode-capacitor cells for PV applications," *IEEE J. Emerg. Sel. Topics Ind. Electron.*, vol. 4, no. 1, pp. 18–27, Jan. 2023, doi: 10.1109/JESTIE.2022.3173178.
- [60] E. Meshkati and H. Farzanehfard, "Family of high step-up multi-input converters with continuous battery current based on three port boost topology and switched capacitors," in *Proc. 14th Power Electron., Drive Syst., Technol. Conf. (PEDSTC)*, Babol, Iran, Jan. 2023, pp. 1–7, doi: 10.1109/PEDSTC57673.2023.10087103.
- [61] S. Franklin, K. Gunasekaran, K. Rajesh, and J. S. Priya, "Multiport converter based solar PV system using flyback converter," in *Proc. 3rd Int. Conf. Artif. Intell. Smart Energy (ICAIS)*, Coimbatore, India, Feb. 2023, pp. 246–250, doi: 10.1109/ICAIS56108.2023.10073790.
- [62] U. Sarma, S. Ganguly, and R. Adda, "Design and control of multi-input single-output DC/DC boost converter for the application of pem fuel cell-battery-hybrid energy system in locomotives," in *Proc. IEEE Power Energy Society General Meeting (PESGM)*, Denver, CO, USA, Jul. 2022, pp. 1–5, doi: 10.1109/PESGM48719.2022.9917089.
- [63] A. Elshora and H. A. Gabbar, "Design and analysis of multi-input, single-output, nonisolated DC/DC converter for fast charging of electric vehicles," in *Proc. Int. Conf. Electr., Comput., Commun. Mechatronics Eng. (ICECCME)*, Mauritius, Mauritius, Oct. 2021, pp. 1–7, doi: 10.1109/ICECCME52200.2021.9591071.
- [64] M. N. Hidayat, A. M. T. Nuban, and F. Ronilaya, "Design and analysis of a single output DC–DC converter with multiple input voltages," in *Proc. Int. Conf. Electr. Inf. Technol. (IEIT)*, Malang, Indonesia, Sep. 2021, pp. 192–198, doi: 10.1109/IEIT53149.2021.9587437.
- [65] Y. Jeong, J.-D. Park, R. Rorrer, K.-W. Kim, and B.-H. Lee, "A novel multi-input and single-output DC/DC converter for small unmanned aerial vehicle," in *Proc. IEEE Appl. Power Electron. Conf. Expo.* (APEC), New Orleans, LA, USA, Mar. 2020, pp. 1302–1308, doi: 10.1109/APEC39645.2020.9124144.
- [66] K. Varesi, A. A. Ghandomi, S. H. Hosseini, M. Sabahi, and E. Babaei, "An improved structure for multi-input high step-up DC–DC converters," in *Proc. 8th Power Electron., Drive Syst. Technol. Conf.* (*PEDSTC*), Mashhad, Iran, Feb. 2017, pp. 241–246, doi: 10.1109/PED-STC.2017.7910330.
- [67] E. Babaei, H. Tarzamni, F. Tahami, H. K. Jahan, and M. B. B. Sharifian, "Multi-input high step-up inverter with soft-switching capability, applicable in photovoltaic systems," *IET Power Electron.*, vol. 13, no. 1, pp. 133–143, Jan. 2020, doi: 10.1049/iet-pel.2018.5801.
- [68] K. Varesi, S. H. Hosseini, M. Sabahi, and E. Babaei, "Modular non-isolated multi-input high step-up DC–DC converter with reduced normalised voltage stress and component count," *IET Power Electron.*, vol. 11, no. 6, pp. 1092–1100, Mar. 2018, doi: 10.1049/iet-pel.2017.0483.
- [69] K. Varesi, S. H. Hosseini, M. Sabahi, E. Babaei, and N. Vosoughi, "Performance and design analysis of an improved non-isolated multiple input buck DC–DC converter," *IET Power Electron.*, vol. 10, no. 9, pp. 1034–1045, Jun. 2017, doi: 10.1049/iet-pel.2016.0750.
- [70] K. Varesi, S. H. Hosseini, M. Sabahi, and E. Babaei, "A high-voltage gain nonisolated noncoupled inductor based multi-input DC–DC topology with reduced number of components for renewable energy systems," *Int. J. Circuit Theory Appl.*, vol. 46, no. 3, pp. 505–518, Nov. 2017, doi: 10.1002/cta.2428.

- [71] A. Affam, Y. M. Buswig, A.-K.-B. H. Othman, N. B. Julai, and O. Qays, "A review of multiple input DC–DC converter topologies linked with hybrid electric vehicles and renewable energy systems," *Renew. Sustain. Energy Rev.*, vol. 135, Jan. 2021, Art. no. 110186, doi: 10.1016/j.rser.2020.110186.
- [72] G. Bayraktar, A. Çabri, and D. Yildirim, "Multi input DC–DC converters for rooftop PV power systems," in *Proc. 11th Int. Conf. Electr. Electron. Eng. (ELECO)*, Bursa, Turkey, Nov. 2019, pp. 230–234, doi: 10.23919/ELECO47770.2019.8990389.
- [73] A. Kumar, N. Bajoria, and A. Alam, "Multiple-input-single-output converter for hybrid renewable sources," in *Proc. Int. Conf. Vis. Towards Emerg. Trends Commun. Netw. (ViTECoN)*, Vellore, India, Mar. 2019, pp. 1–5, doi: 10.1109/ViTECoN.2019.8899579.
- [74] H. AboReada, S. S. Williamson, and V. Sood, "Analysis and control of multi-input, single-output, non-isolated DC/DC converter for effective renewable energy management," in *Proc. IEEE Transp. Electrific. Conf. Expo (ITEC)*, Detroit, MI, USA, Jun. 2019, pp. 1–6, doi: 10.1109/ITEC.2019.8790542.
- [75] R. Ahmed and N. E. Zakzouk, "A single-inductor MISO converter with unified decoupled MPPT algorithm for PV systems undergoing shading conditions," in *Proc. IEEE Int. Conf. Environ. Electr. Eng. IEEE Ind. Commercial Power Syst. Eur.*, Genova, Italy, Jun. 2019, pp. 1–6, doi: 10.1109/EEEIC.2019.8783833.
- [76] M. Dhananjaya and S. Pattnaik, "Design and implementation of a multi-input single-output DC–DC converter," in *Proc. IEEE Int. Conf. Sustain. Energy Technol. Syst. (ICSETS)*, Bhubaneswar, India, Feb. 2019, pp. 194–199, doi: 10.1109/ICSETS.2019.8744815.
- [77] M. Y. Ali Khan, L. Saeed, S. H. Khan, and J. Saleem, "Design of a multi-input single-output DC–DC boost converter for micro grid application," in *Proc. Int. Conf. Eng. Emerg. Technol. (ICEET)*, Lahore, Pakistan, Feb. 2019, pp. 1–6, doi: 10.1109/CEET1.2019. 8711823.
- [78] A. Tomar and S. Mishra, "Grid interactive MISO converter based PV system," in *Proc. 2nd IEEE Int. Conf. Power Electron., Intell. Control Energy Syst. (ICPEICES)*, Delhi, India, Oct. 2018, pp. 642–647, doi: 10.1109/ICPEICES.2018.8897365.
- [79] R. Aravind, B. Chokkalingam, and L. Mihet-Popa, "A transformerless non-isolated multi-port DC–DC converter for hybrid energy applications," *IEEE Access*, vol. 11, pp. 52050–52065, 2023, doi: 10.1109/ACCESS.2023.3280195.
- [80] R. R. Ahrabi, H. Ardi, M. Elmi, and A. Ajami, "A novel step-up multiinput DC–DC converter for hybrid electric vehicles application," *IEEE Trans. Power Electron.*, vol. 32, no. 5, pp. 3549–3561, May 2017, doi: 10.1109/TPEL.2016.2585044.
- [81] M. M. Antony and M. V. Jayan, "A study on a compact single input dual output non-isolated DC–DC converter with low switching stresses," in *Proc. Int. Symp. Ind. Electron. Appl. (INDEL)*, Nov. 2020, pp. 1–6, doi: 10.1109/INDEL50386.2020.9266210.
- [82] M. Rasouli, M. Mehrasa, A. Ganjavi, M. S. Sadabadi, H. Ghoreishy, and A. Ale Ahmad, "Lyapunov-based control strategy for a single-input dual-output three-level DC/DC converter," *IEEE Trans. Ind. Electron.*, vol. 70, no. 10, pp. 10486–10495, Oct. 2023, doi: 10.1109/TIE.2022. 3217610.
- [83] Z. Dong, X. L. Li, C. K. Tse, and Z. Zhang, "Derivation of single-input dual-output converters with simple control and no cross regulation," *IEEE Trans. Power Electron.*, vol. 35, no. 11, pp. 11930–11941, Nov. 2020, doi: 10.1109/TPEL.2020.2983618.
- [84] A. Ganjavi, H. Ghoreishy, and A. A. Ahmad, "A novel single-input dualoutput three-level DC–DC converter," *IEEE Trans. Ind. Electron.*, vol. 65, no. 10, pp. 8101–8111, Oct. 2018, doi: 10.1109/TIE.2018.2807384.
- [85] S. Khan, K. Varshney, M. Zaid, M. Tariq, J. Ahmad, and Z. Sarwer, "A single input dual output high gain DC–DC converter with reduced voltage stress," in *Proc. 47th Annu. Conf. IEEE Ind. Electron. Soc.*, Toronto, ON, Canada, Oct. 2021, pp. 1–6, doi: 10.1109/IECON48115.2021.9589488.
- [86] S. Sherin P. T. and R. Kumar P., "Analysis and implementation of a single input dual output boost derived hybrid converter," in *Proc. Int. Conf. Futuristic Technol. Control Syst. Renew. Energy (ICFCR)*, Malappuram, India, Sep. 2020, pp. 1–6, doi: 10.1109/icfcr50903.2020.9250001.
- [87] Y. Kinoshita and H. Haga, "A wide range output voltage gain operation with mode transition of single input dual output LLC converter," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Detroit, MI, USA, Oct. 2020, pp. 1180–1185, doi: 10.1109/ECCE44975.2020.9236409.

- [88] R. Troudi, S. Moreau, G. Champenois, M. Bouzid, and K. Jelassi, "A new single-input dual-output DC–DC converter modelled using smallsignal AC approach for micro-grid applications," in *Proc. 10th Int. Renew. Energy Congr. (IREC)*, Sousse, Tunisia, Mar. 2019, pp. 1–6, doi: 10.1109/irec.2019.8754581.
- [89] S. Nath, "Effect of mutual coupling on CCM/DCM boundary in single input dual output boost converter," in *Proc. IEEE 8th Power India Int. Conf. (PIICON)*, Kurukshetra, India, Dec. 2018, pp. 1–6, doi: 10.1109/POWERI.2018.8704461.
- [90] M. S. Malik, H. A. Khan, and N. A. Zaffar, "Evaluation of a single inductor based single-input dual-output buck converter for DC microgrid applications," in *Proc. IEEE 7th World Conf. Photovoltaic Energy Convers.* (WCPEC), Jun. 2018, pp. 0613–0617, doi: 10.1109/PVSC.2018.8547281.
- [91] L. Wang, D. S. Yu, R. Xu, Z. B. Ye, H. H. C. Iu, and T. Fernando, "A pulse train controlled single-input dual-output buck converter," in *Proc. IEEE Int. Symp. Circuits Syst. (ISCAS)*, Florence, Italy, May 2018, pp. 1–5, doi: 10.1109/ISCAS.2018.8351803.
- [92] S. Zhou, G. Zhou, X. Liu, and H. Zhao, "Dynamic freewheeling control for SIDO buck converter with fast transient performance, minimized cross-regulation, and high efficiency," *IEEE Trans. Ind. Electron.*, vol. 70, no. 2, pp. 1467–1477, Feb. 2023, doi: 10.1109/TIE.2022.3156169.
- [93] L. Senapati, M.-M. Garg, A. K. Panda, and S. K. Mazumder, "Decoupled voltage mode control of SIDO cuk converter for EV auxiliary power supply," in *Proc. 2nd Int. Conf. Power, Control Comput. Technol. (ICPCT)*, Raipur, India, Mar. 2022, pp. 1–6, doi: 10.1109/ICPC2T53885.2022.9776713.
- [94] M. Dhananjaya, D. Potnuru, P. Manoharan, and H. H. Alhelou, "Design and implementation of single-input-multi-output DC–DC converter topology for auxiliary power modules of electric vehicle," *IEEE Access*, vol. 10, pp. 76975–76989, 2022, doi: 10.1109/ACCESS.2022.3192738.
- [95] S. Markkassery, A. Saradagi, A. D. Mahindrakar, N. Lakshminarasamma, and R. Pasumarthy, "Modeling, design and control of non-isolated single-input multi-output zeta–buck–boost converter," *IEEE Trans. Ind. Appl.*, vol. 56, no. 4, pp. 3904–3918, Jul. 2020, doi: 10.1109/TIA.2020.2984190.
- [96] H. Gupta, "Topologies of single-input, multiple-output DC–DC converters: Design and applications," in *Proc. 6th Int. Conf. Electron., Commun. Aerosp. Technol.*, Coimbatore, India, Dec. 2022, pp. 321–325, doi: 10.1109/ICECA55336.2022.10009405.
- [97] M. S. Jabbar and A. A. Ridha, "Design of single input multi-output DC converter to feed various DC loads," in *Proc. 5th Int. Conf. Eng. Technol. Appl. (IICETA*), Al-Najaf, Iraq, May 2022, pp. 170–175, doi: 10.1109/IICETA54559.2022.9888352.
- [98] R. Arulmozhiyal, M. Murali, and P. Thirumalini, "Single input multi-output DC–DC converter for sustinable energy applications," in *Proc. 2nd Int. Conf. Adv. Electr., Comput., Commun. Sustain. Technol. (ICAECT)*, Bhilai, India, Apr. 2022, pp. 1–5, doi: 10.1109/ICAECT54875.2022.9807941.
- [99] Z. Lin, H. Deng, X. Zhang, H. H. Iu, T. Fernando, C. Townsend, and Y. Liu, "A novel single-input multi-output DC/DC converter with constant current mode operation," in *Proc. 31st Australas. Universities Power Eng. Conf. (AUPEC)*, Perth, Australia, Sep. 2021, pp. 1–5, doi: 10.1109/AUPEC52110.2021.9597812.
- [100] M. Ramesh, B. Mallikarjuna, and T. Rajasekar, "A novel investigation on single-input three-output DC–DC buck converter for electrical vehicles," in *Proc. 7th Int. Conf. Electr. Energy Syst. (ICEES)*, Chennai, India, Feb. 2021, pp. 141–146, doi: 10.1109/ICEES51510.2021.9383635.
- [101] Z. Zhou, N. Tang, B. Nguyen, W. Hong, P. P. Pande, and D. Heo, "A wide output voltage range single-input-multi-output hybrid DC–DC converter achieving 87.5% peak efficiency with a fast response time and low cross regulation for DVFS applications," in *Proc. IEEE Custom Integr. Circuits Conf. (CICC)*, Boston, MA, USA, Mar. 2020, pp. 1–4, doi: 10.1109/CICC48029.2020.9075892.
- [102] P. Priyanka, S. Sathyan, and M. Sahoo, "Current-fed integrated singleinput multi-output (SIMO) switched converter," in *Proc. 15th IEEE India Council Int. Conf. (INDICON)*, Coimbatore, India, Dec. 2018, pp. 1–6, doi: 10.1109/INDICON45594.2018.8987078.
- [103] S. Markkassery, A. D. Mahindrakar, N. Lakshminarasamma, and R. Pasumarthy, "Modelling of non-isolated single-input-multi-output DC-DC converter," in *Proc. IEEE Int. Conf. Power Electron., Drives Energy Syst. (PEDES)*, Chennai, India, Dec. 2018, pp. 1–6, doi: 10.1109/PEDES.2018.8707836.

- [104] R. Gopalasami and B. Chokkalingam, "A photovoltaic-powered modified multiport converter for an EV charger with bidirectional and grid connected capability assist PV2V, G2V, and V2G," *World Electric Vehicle J.*, vol. 15, no. 1, p. 31, Jan. 2024, doi: 10.3390/wevj15010031.
- [105] A. T. L. Lee, W. Jin, S.-C. Tan, and S. Y. Hui, "Single-inductor multipleoutput (SIMO) buck hybrid converter for simultaneous wireless and wired power transfer," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 10, no. 2, pp. 2163–2177, Apr. 2022, doi: 10.1109/JESTPE.2020.3002987.
- [106] R. Gopalasami, B. Chokkalingam, and S. Muthusamy, "A novel method for hybridization of super lift luo converter and boost converter for electric vehicle charging applications," *Energy Sour., A, Recovery, Utilization, Environ. Effects*, vol. 45, no. 3, pp. 8419–8437, Jun. 2023, doi: 10.1080/15567036.2023.2226104.
- [107] Y. Jiang, S. Asar, M. Ahmed, H. Zhang, and A. Fayed, "Output control techniques for dual-frequency SIMO buck converters," *IEEE Trans. Circuits Syst. I, Reg. Papers*, vol. 66, no. 10, pp. 4055–4067, Oct. 2019, doi: 10.1109/TCSI.2019.2922386.
- [108] A. Salimath, E. Botti, G. Gonano, P. Cacciagrano, D. L. Brambilla, T. Barbieri, F. Maloberti, and E. Bonizzoni, "An 86% efficiency, wide-V_{in} SIMO DC–DC converter embedded in a car-radio IC," *IEEE Trans. Circuits Syst. I, Reg. Papers*, vol. 66, no. 9, pp. 3598–3609, Sep. 2019, doi: 10.1109/TCSI.2019.2928310.
- [109] S.-U. Shin, "An adaptively controlled 10-output reconfigurable dual-bus SIMO system with group allocator for diversified load condition," *IEEE Access*, vol. 7, pp. 45440–45450, 2019, doi: 10.1109/ACCESS.2019.2908212.
- [110] N.-S. Pham, T. Yoo, T. T. Kim, C.-G. Lee, and K.-H. Baek, "A 0.016 mV/mA cross-regulation 5-output SIMO DC–DC buck converter using output-voltage-aware charge control scheme," *IEEE Trans. Power Electron.*, vol. 33, no. 11, pp. 9619–9630, Nov. 2018, doi: 10.1109/TPEL.2017.2785838.
- [111] D. Li and G. Chen, "Cross regulation limit exploration of SIMO DC–DC converter using differential evolution algorithm," in *Proc. IEEE Int. Power Electron. Appl. Conf. Expo. (PEAC)*, Guangzhou, China, Nov. 2022, pp. 11–15, doi: 10.1109/PEAC56338.2022.9959228.
- [112] C. Keerthika, S. Ramprasath, P. Rameshbabu, and C. Krishnakumar, "Simulation and steady state analysis of SIMO boost converter for electric vehicles," in *Proc. 2nd Int. Conf. Emerg. Technol. (INCET)*, Belagavi, India, May 2021, pp. 1–5, doi: 10.1109/INCET51464.2021.9456358.
- [113] H. F. Chinchero and J. M. Alonso, "Review on DC–DC SIMO converters with parallel configuration for LED lighting control," in *Proc. IEEE ANDESCON*, Quito, Ecuador, Oct. 2020, pp. 1–7, doi: 10.1109/ANDESCON50619.2020.9272083.
- [114] Z. H. N. Khan, N. Ahmad, D. Khan, H. Abbasizadeh, S. A. A. Shah, Y. J. Park, and K.-Y. Lee, "A SIMO DC–DC boost converter with high efficiency and small area," in *Proc. Int. Conf. Electron., Inf., Commun.* (*ICEIC*), Honolulu, HI, USA, Jan. 2018, pp. 1–3, doi: 10.23919/ELIN-FOCOM.2018.8330673.
- [115] M. Dhananjaya, D. Ponuru, T. S. Babu, B. Aljafari, and H. H. Alhelou, "A new multi-output DC–DC converter for electric vehicle application," *IEEE Access*, vol. 10, pp. 19072–19082, 2022, doi: 10.1109/ACCESS.2022.3151128.
- [116] K. Suresh, N. Chellammal, C. Bharatiraja, P. Sanjeevikumar, F. Blaabjerg, and J. B. H. Nielsen, "Cost-efficient nonisolated three-port DC–DC converter for EV/HEV applications with energy storage," *Int. Trans. Electr. Energy Syst.*, vol. 29, no. 10, Oct. 2019, doi: 10.1002/2050-7038.12088.
- [117] Z. Saadatizadeh, P. C. Heris, and H. A. Mantooth, "Modular expandable multiinput multioutput (MIMO) high step-up transformerless DC-DC converter," *IEEE Access*, vol. 10, pp. 53124–53142, 2022, doi: 10.1109/ACCESS.2022.3175876.
- [118] L. Mo, J. Huang, G. Chen, X. Qing, and Y. Hu, "Computer-aided systematic topology derivation of single-inductor multi-input multioutput converters from working principle," *IEEE Trans. Circuits Syst. I, Reg. Papers*, vol. 69, no. 6, pp. 2637–2649, Jun. 2022, doi: 10.1109/TCSI.2022.3159718.
- [119] J.-D. Chen, K.-W. Wu, H.-L. Ho, C.-T. Lee, and F.-Y. Lin, "3-D multiinput multi-output (MIMO) pulsed chaos LiDAR based on time-division multiplexing," *IEEE J. Sel. Topics Quantum Electron.*, vol. 28, no. 5, pp. 1–9, Sep. 2022, doi: 10.1109/JSTQE.2022.3150791.
- [120] Y. Zhang, G. Spiazzi, S. Buso, and T. Caldognetto, "MIMO control of a high-step-up isolated bidirectional DC–DC converter," *IEEE Trans. Ind. Electron.*, vol. 69, no. 5, pp. 4687–4696, May 2022, doi: 10.1109/TIE.2021.3078393.

- [121] Z. Shan, X. Ding, J. Jatskevich, and C. K. Tse, "Synthesis of multi-input multi-output DC/DC converters without energy buffer stages," *IEEE Trans. Circuits Syst. II, Exp. Briefs*, vol. 68, no. 2, pp. 712–716, Feb. 2021, doi: 10.1109/TCSII.2020.3015388.
- [122] Y. Chen, P. Wang, Y. Elasser, and M. Chen, "Multicell reconfigurable multi-input multi-output energy router architecture," *IEEE Trans. Power Electron.*, vol. 35, no. 12, pp. 13210–13224, Dec. 2020, doi: 10.1109/TPEL.2020.2996199.
- [123] M. Salimi, F. Radmand, and M. H. Firouz, "Dynamic modeling and closed-loop control of hybrid grid-connected renewable energy system with multi-input multi-output controller," *J. Modern Power Syst. Clean Energy*, vol. 9, no. 1, pp. 94–103, Jan. 2021, doi: 10.35833/MPCE.2018.000353.
- [124] X. Wu, Z. Du, X. Yuan, G. Wu, and F. Zeng, "Subsynchronous oscillation analysis of grid-connected converter based on MIMO transfer functions," *IEEE Access*, vol. 8, pp. 92089–92097, 2020, doi: 10.1109/ACCESS.2020.2994219.
- [125] X. L. Li, Z. Dong, C. K. Tse, and D. D. Lu, "Single-inductor multiinput multi-output DC–DC converter with high flexibility and simple control," *IEEE Trans. Power Electron.*, vol. 35, no. 12, pp. 13104–13114, Dec. 2020, doi: 10.1109/TPEL.2020.2991353.
- [126] P. Mohseni, S. H. Hosseini, M. Sabahi, T. Jalilzadeh, and M. Maalandish, "A new high step-up multi-input multi-output DC–DC converter," *IEEE Trans. Ind. Electron.*, vol. 66, no. 7, pp. 5197–5208, Jul. 2019, doi: 10.1109/TIE.2018.2868281.
- [127] M. Jabbari and M. S. Dorcheh, "Resonant multi-input/multioutput/bidirectional ZCS step-down DC–DC converter with systematic synthesis for point-to-point power routing," *IEEE Trans. Power Electron.*, vol. 33, no. 7, pp. 6024–6032, Jul. 2018, doi: 10.1109/TPEL.2017.2749326.
- [128] A. Trivedi and M. Veerachary, "MIMO controller design for two-input DC–DC converter without de-couplers," in *Proc. IEEE IAS Global Conf. Renew. Energy Hydrogen Technol. (GlobConHT)*, Male, Maldives, Mar. 2023, pp. 1–5, doi: 10.1109/GlobConHT56829.2023.10087549.
- [129] V. Bhole, M. Bagewadi, and S. Dambhare, "Multi-input multioutput hybrid converter for remote islanded application," in *Proc. Int. Conf. Electr., Comput., Commun. Mechatronics Eng. (ICEC-CME)*, Maldives, Maldives, Nov. 2022, pp. 1–6, doi: 10.1109/ICEC-CME55909.2022.9988153.
- [130] F. B. Gurbuz, K. Kayisli, S. Demirbas, R. Bayindir, I. Colak, and M. Roscia, "Multi input-multi output (MIMO) converter system fed by wind energy," in *Proc. 11th Int. Conf. Renew. Energy Res. Appl.* (*ICRERA*), Istanbul, Turkey, Sep. 2022, pp. 179–185, doi: 10.1109/ICR-ERA55966.2022.9922847.
- [131] A. Shoaei, K. Abbaszadeh, and H. Allahyari, "A robust hysteresisfeedforward control approach with high flexibility for a singleinductor multi-port DC–DC converter," in *Proc. 30th Int. Conf. Electr. Eng. (ICEE)*, Tehran, Iran, May 2022, pp. 762–768, doi: 10.1109/ICEE55646.2022.9827044.
- [132] A. Asadi, K. Abbaszadeh, and A. Darabi, "Modeling and control of a single-inductor multi-input multi-output DC–DC boost converter for PV applications," in *Proc. 13th Power Electron., Drive Syst., Technol. Conf. (PEDSTC)*, Tehran, Iran, Feb. 2022, pp. 51–55, doi: 10.1109/PED-STC53976.2022.9767320.
- [133] A. A. Asl, R. Alizadeh Asl, and S. H. Hosseini, "A new extendable multi-input multi-output DC–DC converter suitable for renewable energies," in *Proc. 13th Power Electron., Drive Syst., Technol. Conf.* (*PEDSTC*), Tehran, Iran, Feb. 2022, pp. 46–50, doi: 10.1109/PED-STC53976.2022.9767480.
- [134] R. Li, X. Zhang, H. Iu, T. K. Chau, and Y. Liu, "Closed-loop control design for a new multi-input multi-output DC/DC converter," in *Proc.* 31st Australas. Universities Power Eng. Conf. (AUPEC), Perth, Australia, Sep. 2021, pp. 1–5, doi: 10.1109/AUPEC52110.2021.9597788.
- [135] S. Upadhyaya and M. Veerachary, "Interaction quantification in multi-input multi-output integrated DC–DC converters," in *Proc. IEEE 4th Int. Conf. Comput., Power Commun. Technol.* (GUCON), Kuala Lumpur, Malaysia, Sep. 2021, pp. 1–6, doi: 10.1109/GUCON50781.2021.9573935.
- [136] S. Karthikeyan, M. Mouli, P. Nigila, T. Pavithra, and R. Ramya, "Hardware design of multi input and multi output MIMO converter," in *Proc. 7th Int. Conf. Adv. Comput. Commun. Syst.* (*ICACCS*), vol. 1, Coimbatore, India, Mar. 2021, pp. 648–651, doi: 10.1109/ICACCS51430.2021.9442009.
- VOLUME 12, 2024

- [137] F. Mumtaz, N. Z. Yahaya, S. T. Meraj, R. Kannan, B. S. M. Singh, and O. Ibrahim, "Multi-input multi-output DC–DC converter network for hybrid renewable energy applications," in *Proc. Int. Conf. Innov. Intell. Informat., Comput. Technol. (ICT)*, Sakheer, Bahrain, Dec. 2020, pp. 1–6, doi: 10.1109/3ICT51146.2020.9312026.
- [138] X. L. Li, C. K. Tse, and D. D-C. Lu, "Single-inductor multi-input multioutput DC–DC converter with high flexibility and simple control," in *Proc. IEEE Int. Symp. Circuits Syst. (ISCAS)*, Seville, Spain, Oct. 2020, pp. 1–5, doi: 10.1109/ISCAS45731.2020.9180788.
- [139] B. Wang, L. Xian, V. R. K. Kanamarlapudi, K. J. Tseng, A. Ukil, and H. B. Gooi, "A digital method of power-sharing and cross-regulation suppression for single-inductor multiple-input multiple-output DC–DC converter," *IEEE Trans. Ind. Electron.*, vol. 64, no. 4, pp. 2836–2847, Apr. 2017, doi: 10.1109/TIE.2016.2631438.
- [140] A. A. Nilangekar and R. G. Kale, "Design and development of electric vehicle battery charging using MIMO boost converter," in *Proc. Online Int. Conf. Green Eng. Technol. (IC-GET)*, Coimbatore, India, Nov. 2016, pp. 1–6, doi: 10.1109/GET.2016.7916805.
- [141] R. G. Kale and A. A. Nilangekar, "Implementation of multiple input and multiple output boost converter for electric vehicle charging system," in *Proc. Int. Conf. Nascent Technol. Eng. (ICNTE)*, Vashi, India, 2017, pp. 1–6, doi: 10.1109/ICNTE.2017.7947940.
- [142] E. Babaei and O. Abbasi, "Structure for multi-input multi-output DC–DC boost converter," *IET Power Electron.*, vol. 9, no. 1, pp. 9–19, Jan. 2016, doi: 10.1049/iet-pel.2014.0985.
- [143] E. Babaei and O. Abbasi, "A new topology for bidirectional multiinput multi-output buck direct current-direct current converter," *Int. Trans. Electr. Energy Syst.*, vol. 27, no. 2, p. e2254, Feb. 2017, doi: 10.1002/etep.2254.
- [144] M. Alzgool and H. Nouri, "Design, control and modelling of a novel multi-input-multi-output boost converter hub," in *Proc. 10th Jordanian Int. Electr. Electron. Eng. Conf. (JIEEEC)*, Amman, Jordan, May 2017, pp. 1–7, doi: 10.1109/JIEEEC.2017.8051399.
- [145] G. Chen, Z. Jin, Y. Deng, X. He, and X. Qing, "Principle and topology synthesis of integrated single-input dual-output and dual-input singleoutput DC–DC converters," *IEEE Trans. Ind. Electron.*, vol. 65, no. 5, pp. 3815–3825, May 2018, doi: 10.1109/TIE.2017.2760856.
- [146] A. Nahavandi, M. T. Hagh, M. B. B. Sharifian, and S. Danyali, "A nonisolated multiinput multioutput DC–DC boost converter for electric vehicle applications," *IEEE Trans. Power Electron.*, vol. 30, no. 4, pp. 1818–1835, Apr. 2015, doi: 10.1109/TPEL.2014.2325830.
- [147] R. Faraji, L. Ding, M. Esteki, N. Mazloum, and S. A. Khajehoddin, "Soft-switched single inductor single stage multiport bidirectional power converter for hybrid energy systems," *IEEE Trans. Power Electron.*, vol. 36, no. 10, pp. 11298–11315, Oct. 2021, doi: 10.1109/TPEL.2021. 3074378.
- [148] H. Tarzamni, H. S. Gohari, M. Sabahi, and J. Kyyrä, "Nonisolated high step-up DC–DC converters: Comparative review and metrics applicability," *IEEE Trans. Power Electron.*, vol. 39, no. 1, pp. 582–625, Jan. 2024, doi: 10.1109/TPEL.2023.3264172.
- [149] P. Kolahian, H. Tarzamni, A. Nikafrooz, and M. Hamzeh, "Multi-port DC–DC converter for bipolar medium voltage DC micro-grid applications," *IET Power Electron.*, vol. 12, no. 7, pp. 1841–1849, Jun. 2019, doi: 10.1049/iet-pel.2018.6031.
- [150] M. Maalandish, S. H. Hosseini, S. Ghasemzadeh, E. Babaei, and T. Jalilzadeh, "A novel multiphase high step-up DC/DC boost converter with lower losses on semiconductors," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 7, no. 1, pp. 541–554, Mar. 2019, doi: 10.1109/JESTPE.2018.2830510.
- [151] M. H. Qais, H. M. Hasanien, and S. Alghuwainem, "Enhanced whale optimization algorithm for maximum power point tracking of variablespeed wind generators," *Appl. Soft Comput.*, vol. 86, Jan. 2020, Art. no. 105937, doi: 10.1016/j.asoc.2019.105937.
- [152] M. Qais, H. M. Hasanien, and S. Alghuwainem, "Salp swarm algorithmbased TS-FLCs for MPPT and fault ride-through capability enhancement of wind generators," *ISA Trans.*, vol. 101, pp. 211–224, Jun. 2020, doi: 10.1016/j.isatra.2020.01.018.
- [153] H. Y. Mahmoud, H. M. Hasanien, A. H. Besheer, and A. Y. Abdelaziz, "Hybrid cuckoo search algorithm and grey wolf optimiser-based optimal control strategy for performance enhancement of HVDC-based offshore wind farms," *IET Gener, Transmiss. Distrib.*, vol. 14, no. 10, pp. 1902–1911, May 2020, doi: 10.1049/iet-gtd.2019.0801.

- [154] Y. Naderi, S. H. Hosseini, S. G. Zadeh, M. Savaghebi, M. Dahidah, and J. M. Guerrero, "Multi-objective model predictive control for microgrid applications," *Int. J. Electr. Power Energy Syst.*, vol. 154, Dec. 2023, Art. no. 109441, doi: 10.1016/j.ijepes.2023.109441.
- [155] Q. Wei, B. Wu, D. Xu, and N. R. Zargari, "Model predictive control of capacitor voltage balancing for cascaded modular DC–DC converters," *IEEE Trans. Power Electron.*, vol. 32, no. 1, pp. 752–761, Jan. 2017, doi: 10.1109/TPEL.2016.2530869.
- [156] Y. Shan, J. Hu, K. W. Chan, Q. Fu, and J. M. Guerrero, "Model predictive control of bidirectional DC–DC converters and AC/DC interlinking converters—A new control method for PV-wind-battery microgrids," *IEEE Trans. Sustain. Energy*, vol. 10, no. 4, pp. 1823–1833, Oct. 2019, doi: 10.1109/TSTE.2018.2873390.
- [157] J. Hu, Y. Shan, K. W. Cheng, and S. Islam, "Overview of power converter control in microgrids—Challenges, advances, and future trends," *IEEE Trans. Power Electron.*, vol. 37, no. 8, pp. 9907–9922, Aug. 2022, doi: 10.1109/TPEL.2022.3159828.
- [158] J. Hu, Y. Shan, J. M. Guerrero, A. Ioinovici, K. W. Chan, and J. Rodriguez, "Model predictive control of microgrids—An overview," *Renew. Sustain. Energy Rev.*, vol. 136, Feb. 2021, Art. no. 110422, doi: 10.1016/j.rser.2020.110422.
- [159] F. Z. U. Abideen, H. A. Khalid, M. S. Khan, H. Rehman, and A. Hasan, "Direct model predictive control of fuel cell and ultra-capacitor based hybrid electric vehicle," *IEEE Access*, vol. 12, pp. 46774–46784, 2024, doi: 10.1109/ACCESS.2024.3381219.
- [160] H. Zheng, W. Chen, and F. Xie, "Control simulation of flappingwing micro aerial vehicle based on multi-level optimization model predictive control," *IEEE Access*, vol. 12, pp. 40700–40709, 2024, doi: 10.1109/ACCESS.2024.3376646.
- [161] B. Hekimoglu, S. Ekinci, and S. Kaya, "Optimal PID controller design of DC–DC buck converter using whale optimization algorithm," in *Proc. Int. Conf. Artif. Intell. Data Process. (IDAP)*, Malatya, Turkey, Sep. 2018, pp. 1–6, doi: 10.1109/IDAP.2018.8620833.
- [162] A. Alhejji and M. I. Mosaad, "Performance enhancement of gridconnected PV systems using adaptive reference PI controller," *Ain Shams Eng. J.*, vol. 12, no. 1, pp. 541–554, Mar. 2021, doi: 10.1016/j.asej.2020.08.006.
- [163] O. Ibrahim, N. Z. Yahaya, and N. Saad, "State-space modelling and digital controller design for DC–DC converter," *Telkomnika*, vol. 14, no. 2, p. 497, Jun. 2016, doi: 10.12928/telkomnika.v14i2.3042.
- [164] J. Freytes, G. Bergna, J. A. Suul, S. D'Arco, H. Saad, and X. Guillaud, "State-space modelling with steady-state time invariant representation of energy based controllers for modular multilevel converters," in *Proc. IEEE Manchester PowerTech*, Manchester, U.K., Jun. 2017, pp. 1–7, doi: 10.1109/PTC.2017.7981011.
- [165] V. Utkin, A. Poznyak, Y. Orlov, and A. Polyakov, "Conventional and high order sliding mode control," *J. Franklin Inst.*, vol. 357, no. 15, pp. 10244–10261, Oct. 2020, doi: 10.1016/j.jfranklin.2020.06.018.
- [166] H. Fatoorehchi and S. A. Ghorbanian, "Sliding mode control for heartbeat electrocardiogram tracking problem," *J. Chem. Petroleum Eng.*, vol. 53, no. 2, pp. 265–272, 2019, doi: 10.22059/jchpe.2019.286911.1292.
- [167] M. R. Mostafa, N. H. Saad, and A. A. El-Sattar, "Tracking the maximum power point of PV array by sliding mode control method," *Ain Shams Eng. J.*, vol. 11, no. 1, pp. 119–131, Mar. 2020, doi: 10.1016/j.asej.2019.09.003.
- [168] Q. A. Tarbosh, Ö. Aydogdu, N. Farah, M. H. N. Talib, A. Salh, N. Çankaya, F. A. Omar, and A. Durdu, "Review and investigation of simplified rules fuzzy logic speed controller of high performance induction motor drives," *IEEE Access*, vol. 8, pp. 49377–49394, 2020, doi: 10.1109/ACCESS.2020.2977115.
- [169] S. A. Hassan and S. Iqbal, "Automatic car braking system using fuzzy logic controller with environmental factors," in *Proc. 22nd Int. Multitopic Conf. (INMIC)*, Islamabad, Pakistan, Nov. 2019, pp. 1–8, doi: 10.1109/INMIC48123.2019.9022773.
- [170] R. Araria, A. Berkani, K. Negadi, F. Marignetti, and M. Boudiaf, "Performance analysis of DC–DC converter and DTC based fuzzy logic control for power management in electric vehicle application," *J. Européen des Sys. Automatisés*, vol. 53, no. 1, pp. 1–9, Feb. 2020, doi: 10.18280/jesa.530101.
- [171] X. Xiang, C. Yu, L. Lapierre, J. Zhang, and Q. Zhang, "Survey on fuzzy-logic-based guidance and control of marine surface vehicles and underwater vehicles," *Int. J. Fuzzy Syst.*, vol. 20, no. 2, pp. 572–586, Feb. 2018, doi: 10.1007/s40815-017-0401-3.

- [172] K. Loukil, H. Abbes, H. Abid, M. Abid, and A. Toumi, "Design and implementation of reconfigurable MPPT fuzzy controller for photovoltaic systems," *Ain Shams Eng. J.*, vol. 11, no. 2, pp. 319–328, Jun. 2020, doi: 10.1016/j.asej.2019.10.002.
- [173] A. K. D. Velayudhan, "Design of a supervisory fuzzy logic controller for monitoring the inflow and purging of gas through lift bags for a safe and viable salvaging operation," *Ocean Eng.*, vol. 171, pp. 193–201, Jan. 2019, doi: 10.1016/j.oceaneng.2018.10.049.
- [174] M. A. Soliman, H. M. Hasanien, H. Z. Azazi, E. E. El-Kholy, and S. A. Mahmoud, "An adaptive fuzzy logic control strategy for performance enhancement of a grid-connected PMSG-based wind turbine," *IEEE Trans. Ind. Informat.*, vol. 15, no. 6, pp. 3163–3173, Jun. 2019, doi: 10.1109/TII.2018.2875922.
- [175] P. Hema Rani, S. Navasree, S. George, and S. Ashok, "Fuzzy logic supervisory controller for multi-input non-isolated DC to DC converter connected to DC grid," *Int. J. Electr. Power Energy Syst.*, vol. 112, pp. 49–60, Nov. 2019, doi: 10.1016/j.ijepes.2019.04.018.
- [176] M. Leso, J. Žilková, M. Biros, and P. Talian, "Survey of control methods for DC–DC converters," *Acta Electrotechnica et Inf.*, vol. 18, no. 3, pp. 41–46, Sep. 2018, doi: 10.15546/aeei-2018-0024.
- [177] A. K. Sharma, A. Vyas, and P. Sharma, "Comparative study of DC–DC converter with different control techniques," *SSRN Electron. J.*, Mar. 2021, doi: 10.2139/ssrn.3808564.
- [178] J. Javidan, "Design of a ZVS PWM DC–DC converter for high gain applications," *Int. J. Circuit Theory Appl.*, vol. 44, no. 5, pp. 977–995, May 2016, doi: 10.1002/cta.2117.
- [179] C. A. Ramos-Paja, D. G. Montoya, and J. D. Bastidas-Rodriguez, "Sliding-mode control of distributed maximum power point tracking converters featuring overvoltage protection," *Energies*, vol. 11, no. 9, p. 2220, Aug. 2018, doi: 10.3390/en11092220.
- [180] J. M. Riquelme-Dominguez and S. Martinez, "Systematic evaluation of photovoltaic MPPT algorithms using state-space models under different dynamic test procedures," *IEEE Access*, vol. 10, pp. 45772–45783, 2022, doi: 10.1109/ACCESS.2022.3170714.
- [181] N. Swaminathan, N. Lakshminarasamma, and Y. Cao, "A fixed zone perturb and observe MPPT technique for a standalone distributed PV system," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 10, no. 1, pp. 361–374, Feb. 2022, doi: 10.1109/JESTPE.2021. 3065916.
- [182] S. R. Pendem and S. Mikkili, "Assessment of cross-coupling effects in PV String-integrated-converters with P&O MPPT algorithm under various partial shading patterns," *CSEE J. Power Energy Syst.*, vol. 8, no. 4, pp. 1013–1028, Jul. 2022, doi: 10.17775/CSEEJPES.2019. 03330.
- [183] J. Y. Fam, S. Y. Wong, H. B. M. Basri, M. O. Abdullah, K. B. Lias, and S. Mekhilef, "Predictive maximum power point tracking for proton exchange membrane fuel cell system," *IEEE Access*, vol. 9, pp. 157384–157397, 2021, doi: 10.1109/ACCESS.2021. 3129849.
- [184] T. K. Roy, M. A. H. Pramanik, and S. K. Ghosh, "Design of an integral terminal-based sliding mode controller for PV and BESS-based DC microgrids," *Energy Nexus*, vol. 7, Sep. 2022, Art. no. 100130, doi: 10.1016/j.nexus.2022.100130.
- [185] A. Harrag and S. Messalti, "How fuzzy logic can improve PEM fuel cell MPPT performances?" *Int. J. Hydrogen Energy*, vol. 43, no. 1, pp. 537–550, Jan. 2018, doi: 10.1016/j.ijhydene.2017.04.093.
- [186] S. Ahmadi, S. Abdi, and M. Kakavand, "Maximum power point tracking of a proton exchange membrane fuel cell system using PSO-PID controller," *Int. J. Hydrogen Energy*, vol. 42, no. 32, pp. 20430–20443, Aug. 2017, doi: 10.1016/j.ijhydene.2017.06.208.
- [187] O. Abdel-Rahim and H. Wang, "A new high gain DC–DC converter with model-predictive-control based MPPT technique for photovoltaic systems," *CPSS Trans. Power Electron. Appl.*, vol. 5, no. 2, pp. 191–200, Jun. 2020, doi: 10.24295/CPSSTPEA.2020.00016.
- [188] S. Bhattacharyya, D. S. Kumar P, S. Samanta, and S. Mishra, "Steady output and fast tracking MPPT (SOFT-MPPT) for P&O and InC algorithms," *IEEE Trans. Sustain. Energy*, vol. 12, no. 1, pp. 293–302, Jan. 2021, doi: 10.1109/TSTE.2020.2991768.
- [189] A. K. Gupta, R. K. Pachauri, T. Maity, Y. K. Chauhan, O. P. Mahela, B. Khan, and P. K. Gupta, "Effect of various incremental conductance MPPT methods on the charging of battery load feed by solar panel," *IEEE Access*, vol. 9, pp. 90977–90988, 2021, doi: 10.1109/ACCESS.2021.3091502.

- [190] F. Mumtaz, N. Zaihar Yahaya, S. Tanzim Meraj, B. Singh, R. Kannan, and O. Ibrahim, "Review on non-isolated DC–DC converters and their control techniques for renewable energy applications," *Ain Shams Eng. J.*, vol. 12, no. 4, pp. 3747–3763, Dec. 2021, doi: 10.1016/j.asej.2021.03.022.
- [191] K. V. G. Raghavendra, K. Zeb, A. Muthusamy, T. N. V. Krishna, S. V. S. P. Kumar, D.-H. Kim, M.-S. Kim, H.-G. Cho, and H.-J. Kim, "A comprehensive review of DC–DC converter topologies and modulation strategies with recent advances in solar photovoltaic systems," *Electronics*, vol. 9, no. 1, p. 31, Dec. 2019, doi: 10.3390/electronics9010031.



RAJESH VERMA received the B.E., M.E., and Ph.D. degrees in electronics and communication engineering from MNNIT Allahabad, Prayagraj, in 1994, 2001, and 2011, respectively. He is currently an Associate Professor with the Department of Electrical Engineering, King Khalid University, Abha, Saudi Arabia. He has work experience of more than 20 years of teaching and administration at many reputed institutes in India, including MNNIT Allahabad. He was with the Telecom

Industry, New Delhi, India, for four years. His research interests include computer networks, MAC protocols, wireless and mobile communication systems, sensor networks, peer-to-peer networks, and M-2-M networks.



R. ARAVIND received the Diploma degree from the Adhiparasakthi Polytechnic College, Melmaruvathur, India, in 2007, the B.E. degree from the P.M.R. Institute of Technology, India, in 2010, and the M.Tech. degree from SRM University, Kattankulathur, India, in 2014. He is currently pursuing the Ph.D. degree with the Department of Electrical and Electronics Engineering, SRM Institute of Science and Technology, Kattankulathur Campus, Chennai, in 2020. He was

an Assistant Professor with the Adhi College of Engineering and Technology, Tamil Nadu, India, for two years.



BHARATIRAJA CHOKKALINGAM (Senior Member, IEEE) received the B.E. degree in electrical and electronics engineering from the Kumaraguru College of Engineering, Coimbatore, India, in 2002, the M.E. degree in power electronics and drives from the Government College of Technology, Coimbatore, in 2006, and the Ph.D. degree from the SRM Institute of Science and Technology, in 2015. He held a Postdoctoral Fellowship with the Centre for Energy and Elec-

tric Power, Faculty of Engineering and the Built Environment, Tshwane University of Technology, South Africa, in 2016, and the Department of Electrical and Computer Engineering, Northeastern University, USA. He was a Visiting Research Scientist with Northeastern University, Boston, MA, USA, in 2018, and a Visiting Researcher with the University of South Africa, in 2019 and 2020. He is currently a Professor with the Department of Electrical and Electronics Engineering, SRM Institute of Science and Technology, Kattankulathur Campus, Chennai, India. He is also the Centre Head of the Centre for the E-Mobility, SRM. He is handling/handled government-funded projects total worth Rs. 15.65 Cr, including SERB, DST, and TNSCST. He received the National Research Foundation Funding. He has authored more than 180 research articles, ten book chapters, and ten IPR patents in his name. He also made three technology transfers. He is a Senior Member of IEI and IET. He is also a member, a mentor, and an advisor of many government and private companies bodies. He was a recipient of the DST, Indo-U.S. Bhaskara Advanced Solar Energy, in 2017, and the Young Scientists Fellowship, Tamil Nadu State Council for Science and Technology, in 2018. He was a recipient of the World's Top 2% Scientists 2021 and 2022 in the energy category. For more information: https://www.srmist.edu.in/faculty/dr-c-bharatiraja/.



SAKTHIVEL ARUCHAMY (Member, IEEE) received the Bachelor of Engineering degree in electrical and electronics engineering and the Master of Engineering degree in power electronics and driving from Bharathiar University, India, in 1999 and 2002, respectively, and the Ph.D. degree from Anna University, Chennai, in 2016. He was a Sabbatical with Cape Breton University, Nova Scotia, Sydney, Canada. He is currently a Professor and the HoD of the Department of Electrical

and Electronics Engineering, New Horizon College of Engineering, Bengaluru, Karnataka, India. His research interests include power converters, harmonic mitigation techniques, electric vehicles, power quality, and soft computing techniques. He is a Life Member of Indian Society for Technical Education (ISTE). He was the Verschuren Centre Research Fellow.



LUCIAN MIHET-POPA (Senior Member, IEEE) was born in 1969. He received the bachelor's degree in electrical engineering, the master's degree in electric drives and power electronics, and the Ph.D. and Habilitation degrees in electrical engineering from Politehnica University Timisoara, Romania, in 1999, 2000, 2002, and 2015, respectively. From 1999 to 2016, he was with Politehnica University Timisoara. He was a Research Scientist with Danish Tech-

nical University, from 2011 to 2014, and Aalborg University, Denmark, from 2000 to 2002. He was a Postdoctoral Researcher with Siegen University, Germany, in 2004. Since 2016, he has been a Full Professor of energy technology with Østfold University College, Norway. He is also the Head of the Research Laboratory "Intelligent Control of Energy Conversion and Storage Systems." He has been granted and participated in more than 25 international projects. He has been awarded more than ten national research grants. He has published more than 200 papers in national and international journals and conference proceedings and 15 books. His research interests include modeling, simulation, control, and testing of energy conversion systems; and distributed energy resources (DER components and systems, including battery storage systems for electric vehicles and hybrid cars and vanadium redox batteries), but also interactive buildings in smart grids.

...